

**Delft University of Technology** 

## Wayfinding behaviour in a multi-level building: A comparative study of HMD VR and **Desktop VR**

Feng, Yan; Duives, Dorine C.; Hoogendoorn, Serge P.

DOI 10.1016/j.aei.2021.101475

**Publication date** 2022 **Document Version** Final published version

Published in Advanced Engineering Informatics

### Citation (APA)

Feng, Y., Duives, D. C., & Hoogendoorn, S. P. (2022). Wayfinding behaviour in a multi-level building: A comparative study of HMD VR and Desktop VR. *Advanced Engineering Informatics*, *51*, 1-22. Article 101475. https://doi.org/10.1016/j.aei.2021.101475

#### Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.



Contents lists available at ScienceDirect

## **Advanced Engineering Informatics**



journal homepage: www.elsevier.com/locate/aei

# Wayfinding behaviour in a multi-level building: A comparative study of HMD VR and Desktop VR



## Yan Feng<sup>\*</sup>, Dorine C. Duives, Serge P. Hoogendoorn

Department of Transport & Planning, Delft University of Technology, Delft, the Netherlands

#### ARTICLE INFO

#### ABSTRACT

Keywords: Wayfinding behaviour Route choice Exit choice Desktop VR HMD VR Evacuation Human–computer interaction

Virtual Reality (VR) provides the possibility to study pedestrian wayfinding behaviour in multi-level buildings. Although VR has been applied increasingly to study pedestrian behaviour, it has remained unclear how different VR technology would affect behavioural outcomes in a multi-level building. The study compares the adoption of different VR technologies for pedestrian wayfinding studies, via investigating the difference in pedestrian wayfinding behaviour and user experience. Wayfinding experiments with two groups of participants were conducted using either HMD VR or Desktop VR. Pedestrian movement trajectory data were collected via the VR system and user experience data were recorded via a questionnaire. These data allow for direct comparison and detailed analysis of pedestrian behaviour and user experience between the adoption of two VR technologies. The results showed that technological differences have a significant impact on wayfinding task performance and head rotation change. However, the route choice, exit choice and user experience were overall similar between the two groups. These results provide empirical evidence supporting researchers to choose between immersive and nonimmersive VR when study pedestrian wayfinding behaviour.

#### 1. Introduction

Pedestrians perform wayfinding activities in buildings on a daily basis. Wayfinding here is defined as a decision-making process in which pedestrians determine the route to a destination and finding an exit to leave the building [1]. Performing wayfinding in large-scale and multi-level buildings, such as train stations, hospitals, and shopping malls, can be difficult because of the complexity of the three-dimensional environment [2]. That is, the complexity of finding one's route and exit in multi-level buildings increases by the multiple floor layouts, complex spatial structures, many indoor objects, and moving along vertical distances [3–5]. In case of an emergency, pedestrian route and exit choice are of vital importance to their survival.

Traditionally, field experiments and surveys have been widely used to investigate pedestrian wayfinding behaviour under both normal and emergency situations (e.g., [6–11]). However, these methods have constraints in terms of experimental control, cost, and data accuracy for studying pedestrian wayfinding behaviour [12]. Most experimental studies focused on a single-level or simplified environment with an experimental area of limited size (e.g., [7,13–17]). Consequently, most experimental conditions featured in traditional pedestrian wayfinding studies differ greatly from actual reality where pedestrians are faced with more complex situations. Moreover, the accuracy of behavioural data is highly influenced by the sensor setup and techniques, and it often requires a large investment in labour to extract the collected data [12]. Additionally, it is time-consuming and challenging to obtain approval to perform a field observation or create an artificial experiment environment. Furthermore, there are ethical and financial constraints to create real and stressful situations to provide participants with a strong sense of presence to make them fully participate and keep focused on the task [21]. We suspect that the existing constraints of the traditional data collection methods (partially) induce a lack of studies featuring pedestrian wayfinding behaviour in large-scale and multi-level buildings [12].

To overcome the existing constraints of traditional data collection methods, the usage of Virtual Reality (VR) to investigate pedestrian wayfinding behaviour has become increasingly popular. With VR, it is possible to place participants in complex or hazardous situations that are costly, stressful or even impossible to simulate in the real world (e.g., [16,22–24]). VR allows researchers to perform controlled experiments that have high internal validity due to their experimental design and provide enhanced ecological validity due to the high-fidelity virtual environment [25]. Additionally, it provides the possibility of accurate

https://doi.org/10.1016/j.aei.2021.101475

Received 7 June 2021; Received in revised form 4 November 2021; Accepted 16 November 2021 Available online 27 November 2021

1474-0346/© 2021 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

<sup>\*</sup> Corresponding author. *E-mail address:* y.feng@tudelft.nl (Y. Feng).

tracking and recording a large variety of data pertaining to pedestrian's movement and choice behaviour in complex environments, such as timestamp, pedestrian movement trajectory, head rotation, and eye movement.

Different VR technologies have been used to study pedestrian wayfinding behaviour, such as head-mounted-display (HMD) (e.g., [13,21,23,26–31]), Desktop VR (e.g., [17,32–35]) and cave automatic virtual environment (CAVE) (e.g., [36–38]). Although studies have demonstrated the effectiveness of VR to study pedestrian behaviour in a variety of cases, one unresolved issue is that the suitability of different VR technologies for pedestrian wayfinding behaviour study is still open to debate. Different VR technologies have different characteristics, it may cause people to perceive the virtual environment differently and behave in the virtual environment differently [15].

While VR technologies are more and more readily available to study wayfinding behaviour, researchers have limited insights into the impact of the adopted VR technologies on their research findings. To date, only a few studies have compared pedestrian wayfinding behaviour or evacuation behaviour using different VR technologies. For example, Santos et al. [15] found better performance with Desktop VR compare to HMD VR, Ruddle & Péruch [39] found no difference in wayfinding performance between HMD VR and Desktop VR, Ronchi et al. [37] showed consistent results of evacuation behaviour between a mobile-HMD and CAVE system in a tunnel emergency scenario. However, the above-mentioned studies did not directly compare the presentation of the same virtual environment using different VR technologies on pedestrian wayfinding behaviour across a variety of wayfinding tasks, and their experimental environments were relatively simple (i.e., singlelevel, limited size, and maze layout). In order to identify the influence of VR technology on pedestrian wayfinding behaviour in large-scale and multi-level environments, it is essential to directly compare VR technologies regarding their impact on pedestrian wayfinding behaviour and user experience in one environment.

The objective of this paper is to compare the adoption of different VR technologies (i.e., HMD VR and Desktop VR) for pedestrian wayfinding studies, via investigating the difference in pedestrian wayfinding behaviour and user experience. HMD VR and Desktop VR are two VR technologies that have been increasingly applied to study pedestrian behaviour. Compared to CAVE, which is very costly and requires a large space for the screen monitors or multiple television projection systems [40], HMD VR and Desktop VR provides cost-effective solutions. Moreover, the access to CAVE system is generally restricted to few institution laboratories, while HMD VR and Desktop VR are more accessible and affordable to a wider range of researchers who are interested in using VR. Meanwhile, compare with HMD VR, Desktop VR decreases technical complexity to implement and even provides lower-cost and off-the-shelf alternatives.

In the current study, we conduct VR experiments in which participants use either HMD VR or Desktop VR to perform a set of wayfinding tasks. In particular, pedestrian behaviour data (i.e., three-dimensional movement trajectories, head rotations, and gaze points) and participant's experience using VR (i.e., perceived realism, usability, feeling of presence, and simulation sickness) are collected synthetically. Previous studies from literature argue that the different features of HMD VR and Desktop VR may cause users to perceive the virtual environment differently and behave differently in the virtual environment (e.g., [41–44]). Thus, the collected data were analysed quantitatively to examine whether pedestrian wayfinding behaviour, such as route and exit choice, observation behaviour, and wayfinding task performance as well as user experience are different between HMD VR and Desktop VR. Accordingly, four hypotheses related to pedestrian wayfinding behaviour and user experience are formulated, namely:

H 1. There is a significant difference in route and exit choice behaviour (i.e., wayfinding strategy, paths, decision points,

staircases, and evacuation exit choice) between the participants that adopted Desktop VR and HMD VR.

**H 2**. There is a significant difference in observation behaviour (i.e., head rotation and gaze point) between the participants that adopted Desktop VR and HMD VR.

**H 3**. There is a significant difference in wayfinding task performance (i.e., time, speed, and distance) between the participants that adopted Desktop VR and HMD VR.

**H 4.** There is a significant difference in user experience (i.e., realism, presence, simulation sickness, and usability) between the participants that adopted Desktop VR and HMD VR.

There are three major contributions of this study, namely we (1) investigate pedestrian wayfinding and evacuation behaviour in a complex and multi-level building using VR, (2) provide a direct comparison of pedestrian wayfinding behaviour and user experience between two different VR technologies, and (3) recommend which VR technology to use to perform pedestrian wayfinding behaviour studies.

The paper is organised as follows. Section 2 provides an overview of used metrics to measure wayfinding behaviour and a review of VR - wayfinding studies. Section 3 describes the experimental method. Accordingly, Section 4 reports the results pertaining to pedestrian wayfinding behaviour in the virtual environment and user experience. Based on the results, Section 5 discusses the findings regarding differences in pedestrian wayfinding behaviour and user experience between the usage of HMD and Desktop VR. Finally, Section 6 presents the conclusions and future work of this paper.

#### 2. Related work

The current study focuses on comparing pedestrian wayfinding behaviour in a multi-level building and user experience of the VR technology between the adoption of HMD VR and Desktop VR. Therefore, this section first provides an overview of commonly used metrics to measure pedestrian wayfinding behaviour in previous studies. Second, this section gives a summary of wayfinding studies that employed different VR technologies.

#### 2.1. Wayfinding behaviour in multi-level buildings

Pedestrian wayfinding studies investigate how people orient themselves and navigate from an origin to a destination [45]. The term "wayfinding" was originally introduced by Lynch [46] where he defined human wayfinding as "a consistent use and organization of definite sensory cues from the external environment". Afterwards, multiple disciplines, such as engineering, psychology, and architecture have developed a wide variety of theories to understand this behavioural process. Jamshidi and Pati [47] classifies wayfinding theories into four categories, namely theories of (1) perception, (2) spatial knowledge development, (3) mental representation of spatial knowledge, and (4) spatial cognition. In general, the act of wayfinding can be viewed as a continuous problem-solving process requiring information about the environment [1,48-50], which contains the process of perception, cognition and decision making [47,49]. To be more specific, wayfinding refers to the process that people acquire information regarding their environment through their senses, understand and manipulate this information, establish a plan, transfer this plan into behavioural activities, and execute these activities in the environment [29,49–50].

Wayfinding behaviour has been widely explored in various spatial settings, including urban spaces (e.g., [51–54]) and buildings (e.g., [4,13,16,29,45,55–56]). Everyday people need to find their way in complex and multi-level buildings, such as offices, university buildings, train stations, hospitals, and shopping malls. Previous studies have observed the difficulty (e.g., disorientation, frustration, and stress) of people finding their way in complex and multi-level buildings [57–59]. Wayfinding in multi-level buildings has been considered complex due to

the navigation of multiple floor layouts, (turning) staircases, differences in visual accessibility, and architectural differentiation. In general, literature identifies three levels of metrics to evaluate pedestrian wayfinding behaviour, namely decision making (e.g., wayfinding strategy, route choice, and exit choice), observation behaviour (e.g., head rotation and gaze point), and wayfinding task performance (e.g., time, speed, and distance) [13,56,58,60]. The explanations of these metrics and wayfinding studies in which these metrics are measured are given below.

Regarding the *decision-making level*, the usage of decision points and paths are found to be closely related to route choice and exit choice [29]. That is, if a person chooses a long route between an origin and a destination (exit), the number of decision points and length of path increase [29]. Literature shows that the arrangement of decision points, their linking paths, and the position of staircases contribute prominently to the experienced complexity of buildings [58]. Moreover, Hölscher et al. [61] illustrated that the adopted wayfinding strategy can also influence the efficiency of pedestrian wayfinding in multi-level buildings. Here, each element included at the decision-making level can be operation alized as follows:

- 1. *Decision points* locations where pedestrians have more than one choice of direction to continue the way [45]. Studies showed that the number of decision points is positively related to the difficulty of wayfinding tasks [29,49,62].
- 2. *Staircases* important vertical interconnections between different floors in a mulita-level building. Staircases could be seen as decision points on the vertical level. Literature found that floor changes that involve vertical movement on staircases can cause disorientation and hinder wayfinding performance [48,58–59]
- 3. *Paths* the smallest segment connecting two decision points that people can move along [46,63]. When choosing between available paths, studies found that people prefer paths with longer lines of sight and are wider [64–67].
- 4. *Wayfinding strategy* the strategy that people adopt to identify their optimal path. According to literature, wayfinding strategy can be categorised into three classifications, namely (1) the floor strategy: the individual first moves to the floor of the destination, (2) the direction strategy: the individual first moves to the horizontal position of the destination as directly as possible (irrespective of level-changes), and (3) the central point strategy: the individual finds the way by visiting the well-known parts of the building [61].

Wayfinding performance measures how well participants perform wayfinding tasks [4]. Often, wayfinding performance is measured using either travel time, travel distance, and/or travel speed. Weisman [68] provided four types of environmental elements that influence pedestrian wayfinding performance in buildings, which were investigated by many studies including (1) visual access to see other parts of the building from a given location (e.g., [69-71]), (2) the degree of architectural differentiation, which is the difference between objects in the building (e.g., [24,29,64]), (3) signs and room numbers to provide identification or directional information (e.g., [13,17,65,72-73]), and (4) plan configuration of the building (e.g., [55,58,74]). Therefore, these environmental factors should be taken into account while developing the virtual building for VR studies. Besides the environmental factors, literature has shown that personal factors, such as gender (e.g., [13,29,75]), age (e.g., [76-77]), and familiarity with the environment (e.g., [78-79]) can affect wayfinding performance. Moreover, the influence of interaction with other people on wayfinding performance has also been studied by [22,56,80-81].

Regarding the *observation behaviour*, pedestrians aid their navigation by looking around in the environment during wayfinding [82], and their gaze behaviour can reveal insights into the information acquisition process that supports the wayfinding tasks [66]. With the development of tracking technologies, such as eye-tracking, motion-tracking, it is possible to detect and collect people's head movements and eye movements during their navigation. Collected data can be used to measure how space is perceived and how specific elements attract individual's attention during wayfinding [83–84]. The usage of eye-tracking and head-tracking in wayfinding studies has been studied in real-life scenarios and virtual environments, such as buildings and urban spaces (e. g., [54,66,85–90]).

#### 2.2. Wayfinding studies in VR

Virtual reality (VR) is defined as a system composed of interactive computer simulations that senses participant's position and responds to their movement, thereby giving participants the feeling of being immersed in a virtual environment [91]. Due to the rapid advancements of high-quality simulations and computer processing power, in combination with the reduction of computer power costs, VR has been applied increasingly to study pedestrian wayfinding behaviour (e.g., [21-22,36]). By using VR, researchers can create environments that are suitable for their research objectives with high experimental control, and let participants experience the virtual world through a continuous stream of high-realistic images and sound landscapes. Human performance in the virtual environment is generally influenced by the individual's level of experienced immersion in the virtual environment [92–93]. Regarding the level of immersion, VR studies can be generally categorised into two groups, namely non-immersive VR and immersive VR.

Non-immersive VR utilises common PC monitors or projections to allow participants to view the virtual environment. Participants typically use abstract interfaces (e.g., joystick, mouse, and keyboard) to control their movements. Several studies applied non-immersive VR to study pedestrian wayfinding and evacuation behaviour. For instance, Desktop VR has been used to study pedestrian wayfinding behaviour during evacuations (e.g., [17,32–35]). Projection-based VR has been applied to investigate pedestrian route selection during evacuations (e. g., [64,94]). Generally, using non-immersive VR, participants can still view the real world, which might limit their sense of immersion [95].

Immersive VR usually requires participants to wear a headset that blocks participants from their real-life environment. Participants interact with the virtual environment through specialist simulator control devices (e.g., controllers and gloves) and motion tracking hardware (e.g., eye, head, and motion tracking devices). One type of immersive VR is the Cave Automatic Virtual Environment (CAVE), which displays the virtual environment on huge screen monitors or multiple television projection systems simultaneously [40]. The CAVE has, for instance, been applied to investigate pedestrian wayfinding in high-rise buildings [3], tunnels [19,36,37,81], and train stations [38]. However, the installation of CAVE systems requires large spaces and the cost is relatively high. Therefore, studies using CAVE were conducted by few research groups with the resources for a CAVE [96].

Another type of frequently used immersive simulator is the headmounted display (HMD), which typically features high-resolution displays. A large body of studies used HMD to investigate pedestrian wayfinding behaviour. One major research theme using HMDs focused on the influence of environmental characteristics on pedestrian wayfinding behaviour, amongst other things, the influence of signage [13,21,27–28], lighting conditions [23,29] and visual cues [24,26,30,71]. Another theme of studies focused on the impact of social influence on pedestrian wayfinding behaviour (e.g., [22,31,56,78]). Compared to CAVE, HMD VR can be built with high flexibility and low cost.

As mentioned in Section 2.1, spatial perception and cognition are important elements during wayfinding in real-life environments, which is the process of obtaining information through individual's senses [29]. The same applies to navigation in a virtual environment, where people need to perceive and obtain knowledge from the virtual environment. Various VR technologies are available in today's market. These VR technologies have different characteristics, such as levels of immersion, interaction ability, and in particular, the costs associated with their installation and use for research purposes vary greatly [37]. A number of studies showed that the usage of different VR technologies can have a varying impact on the user, particularly related to their sense of immersion and presence [41–43], usability [97], and motion sickness [98–99]. The differences in experiencing and perceiving the environments may cause people to behave differently during wayfinding and affect their wayfinding performance (e.g., [41–44]). Therefore, it has become important for researchers who are considering the usage of VR for wayfinding study to understand the assets and limitations of each VR technology.

Although studies have illustrated the effectiveness of VR to study pedestrian wayfinding behaviour, very few studies investigated the impact of different VR technologies on the actual pedestrian behaviour of participant's and their user experience. Several studies compared participant's performance of navigation tasks between Desktop VR and HMD VR [14-15,39,82]. Ruddle and Péruch [39] found that there were no differences in wayfinding performance and route knowledge between Desktop VR and HMD VR. At the same time, Ruddle et al. [82] found that people who used HMD VR travel quicker than people who used Desktop VR, while Hsieh et al. [14] found that people find destinations quicker using Desktop VR than HMD VR and Santos et al. [15] found better performance with Desktop VR compare to HMD VR. In these studies, the experimental environments were abstract mazes, and the tasks were relatively simple [100]. More recently, one study used a HMD and a PC screen to compare task performance in a multi-level indoor environment, which showed that the performance of navigation tasks was better in the Desktop VR than in the HMD VR [44]. However, another study compared pedestrian evacuation behaviour using smartphone-based HMD and CAVE and showed the consistency of pedestrian behaviour between the two VR systems [37].

In reality, pedestrians need to find their way in multi-level buildings and their wayfinding behaviour is affected by the layout of the architectural setting and the quality of the environmental information [100]. Simple and abstract environments have a significant lack of detailed environmental elements that aid pedestrian wayfinding, such as architectural differentiation, distinguishable decorations, visual accessibility, and information signage, which are of great importance for people perceiving the environment [68]. Moreover, the multiple floor layouts in multi-level buildings require movements on a vertical space, which further increases the complexity of wayfinding. Since the complexity and difficulty of wayfinding in multi-level buildings increase [50], findings featured in simplified environments cannot be directly generalised to multi-level buildings, which highlight the importance of investigating the differences in pedestrian behavioural outcomes in multi-level buildings and user experience between the adoption of different VR technologies.

In conclusion, VR technologies, especially HMD VR and Desktop VR, have been adopted increasingly to study pedestrian wayfinding behaviour. These VR technologies have very different characteristics regarding their usability and levels of immersion, presence, simulation sickness, which may cause people to perceive the virtual environment differently and behave differently in the virtual environment [41–44]. Currently, the impact of different VR technologies on the behavioural outcomes of the experimental studies, especially pedestrian wayfinding behaviour studies, is undetermined. Moreover, the few studies that compared the behavioural outcomes of different VR technologies featured relatively simple experimental setups. Consequently, the findings of these VR studies cannot be generalised to complex buildings. Thus, up to the moment, no studies directly compared pedestrian wayfinding behaviour in complex multi-level buildings as well as user experience between the adoption of different VR technologies. In our study, we fill this gap by conducting four wayfinding experiments in a multi-level building using both HMD VR and Desktop VR and comparing the resulting wayfinding behaviour and user experience of participants.

#### 3. Materials and method

In the current study, we designed and conducted VR experiments with HMD VR and Desktop VR. The VR experiment was approved by the Human Research Ethics Committee of the Delft University of Technology (Reference ID 944). This section presents a detailed description of the experimental method.

#### 3.1. The virtual environment

The virtual environment featured a virtual building that comprises four floors. Fig. 1 shows the front view of the virtual building. Each floor features two parallel hallways, multiple intersections, four staircases and four elevators. There are five major exits on the ground floor (see Fig. 1). This virtual environment was originally developed as a VR research tool to study pedestrian wayfinding behaviour in a multi-level building (see [101]). The VR tool was developed using Maya and Unreal Engine 4 (UE4), which supports free navigation and collects pedestrian walking trajectories, head rotations, and gaze points automatically.

The virtual building is based on the building of the Faculty of Civil Engineering and Geoscience of the Delft University of Technology. In addition to the overarching geometry (i.e., the internal layout of the building, walls, escalators, and staircases), detailed environmental elements were also included in the virtual environment in order to improve the accuracy of the building's representation and increase its realism. Weisman [68] identified four classes of environmental elements that influence pedestrian wayfinding in buildings, including (1) visual access, which provides views that one can see other parts of the building from a given location, (2) architectural differentiation, which is the difference of objects in the building regarding size, colour, location, etc., (3) signs to provide identification or directional information, and (4) plan configuration of the building [55,58]. To implement these elements, detailed environmental objects were added in the virtual building accordingly, namely (1) glass windows to represent visual access, (2) various furniture such as chairs, cabinets, and tables that represent architectural differentiation, (3) evacuation signs, exit signs, and room numbers represent signs, and (4) floor plan represent plan configuration (see Fig. 2). The colour and texture of these environmental objects in the virtual environment were modelled as close as possible to realistically represent the real-world experience.

In the virtual environment, the participants have a first-person perspective. The participant's viewpoint is represented by a virtual camera. Participants could move in the virtual environment at a maximum constant speed of 140 cm/s. This speed limit was determined based on pilot tests, which to ensure that participants could have a similar walking pace as in real-life environments without causing simulation sickness (e.g., [102–103]).

#### 3.2. Experimental setup

The versions of HMD VR and Desktop VR of the VR research tool were developed for this study. This sub-section introduces the general setup of the experiment and the setup of HMD VR and Desktop VR.

#### 3.2.1. General technique setup

The VR experiment with the HMD group and the Desktop group were both conducted in the same room (4.6 m  $\times$  3.5 m) with a 2.5 m high ceiling, illuminated by fluorescent lighting, with neither reflective surfaces nor exposure to natural lighting. In both groups, the same computer was used, which was equipped with a AMD Ryzen 7 2700X with a 3.7 GHz CPU, MSI NVIDIA GeForce RTX 2080 graphics card, 16 GB system memory, and a Samsung 970 EVO MZ-V7E500BW 500 GB SSD. Participants also wore a pair of over-ear headphones to receive audio information and isolate themselves from the noise of the real-life environment. The software packages used for running the virtual environment were UE4 and SteamVR.



Fig. 1. The front view of the virtual building.



Fig. 2. A screenshot of the details in the virtual building.

3.2.2. Setup of the HMD VR The HTC Vive system was employed for the experiment of the HMD



Fig. 3. A participant using the HMD VR during the experiment.

group, which mainly included one HMD, one wireless hand controller and two base stations. Fig. 3 shows the devices included in the employed system. The HMD has 2160  $\times$  1200 pixels combined resolution (i.e., 1080  $\times$  1200 per eye), a 110-degree field of view, and a 90 Hz refresh rate for both screens.

A combination of an open-world navigation solution and a steering locomotion was adopted in the HMD group, which means participants had continuous movement with 'step-by-step' effects in the virtual environment when slowing down. The steering locomotion method was adopted because it generates less motion sickness compared to the teleportation method during the prototype tests. Moreover, this combination provides greater navigational benefits and is more natural to use than walking on a treadmill [39,104–105]. Participants used one hand controller to move through the virtual environment. By holding the home pad of the controller, the participant moved forward, and by releasing the home pad, the participant stopped moving. The direction of the movement was controlled by the participant's head orientation.

Tracking of the participants' positions and orientations in the virtual environment was achieved by means of the two base stations. These two base stations were placed opposite each other and connected via a sync cable. HTC Vive provides a room-scale technology that allows the user to walk freely, and the HMD features SteamVR Tracking technology which provides 360-degree head-tracking. The participants were tracked within a space of 3.4 m  $\times$  2.5 m in the experimental room.

#### 3.2.3. Set up of the Desktop VR

The participants of the Desktop group viewed the virtual environment via a 24-inch desktop monitor (AOC G2460PF). The monitor is 565.4 mm long and 393.6 mm high. It has  $1920 \times 1080$  resolution, a refresh rate of 144 Hz, and 1 ms response time. The monitor was placed

on top of a rectangular table (90 cm  $\times$  150 cm) in the same experimental room as the HMD group (see Fig. 4). The horizontal distance between the monitor and the participant was approximately 60 cm.

A combination of an open-world navigation solution and a smooth artificial locomotion style was adopted for the Desktop VR. Participants' positions and orientations in the virtual environment were tracked via the virtual camera that represented participants' viewpoints. This combination allows participants to move artificially in the virtual building via the keyboard and mouse [97]. The participant moved forward by means of the keyboard key (i.e., 'w'), and changed the direction of view and movement by rotating the mouse. As a result of this navigation solution, participants' movement in the virtual building is continuous. Moreover, the setup allowed participants to have 360-degree freedom to move on the horizontal level and 360-degree views on both horizontal and vertical levels.

#### 3.3. Experiment design

A single-factor between-subjects experimental design was used for this study to reduce the learning effects because of repetitive exposure. Literature identifies that repetitive exposure to the same environment affects pedestrian wayfinding performance [16,106]. Especially in the current study, the experimental environment and tasks are exactly the same for both groups. The VR experiments featured two different experiment settings but the only difference between both setups is the HMD VR versus the Desktop VR. Half of the participants were pseudorandomly assigned to the HMD group and the other half to the Desktop group to ensure similar gender distributions between the two groups. During the experiment, the participants of both groups were asked to complete four wayfinding tasks during the experiment, including three wayfinding tasks under normal conditions and one wayfinding task under the evacuation condition.

Fig. 5 shows the abstract layout of the experimental environment. First, pedestrian wayfinding behaviour across the horizontal level was investigated. Participants were initially positioned in front of room 4.02 and were asked to find their way from room 4.02 to room 4.99 (see Fig. 5), which ensured they need to cross from one main corridor to the other and walk the length of the building. Second, pedestrian wayfinding behaviour (including staircase choice) at the vertical level was investigated. Participants were asked to find their way from room 4.99 to room 2.01. This task required participants to move between floors and walked the length of the building. Third, pedestrian wayfinding behaviour across both the horizontal and vertical levels was investigated. Participants were asked to find their way from room 2.01 to room 4.64, which forced them to switch floors and main corridors. The fourth task of the experiment was to investigate pedestrian wayfinding and evacuation choice during an evacuation scenario. When participants



Fig. 4. A participant using the Desktop VR during the experiment.

arrived at room 4.64, the evacuation alarm was triggered with a voice message: "Attention, please leave the building using the emergency exits as indicated. Do not use the elevators.". This evacuation alarm is the same alarm sound that is used during the real-life evacuation procedure at the modelled faculty building. Participants were asked to evacuate and find an exit. Once participants arrived at an exit, the experiment ended.

#### 3.4. Experimental procedure

A consistent experimental procedure was used for the HMD group and the Desktop group. The procedure included five major stages:

- 1. Introduction: When participants arrived at the experimental room, they first read the instruction letter about the experiment, including the usage of HMD VR or Desktop VR, the general procedure of the experiment and safety measures in case of any discomfort during the experiment. Participants were also informed that they had the right to stop the experiment at any time.
- 2. Practice: Participants then were instructed to wear the HMD or sit in front of the desktop monitor. During the practice session, participants were asked to find their way from A to B to C in a simple virtual scenario. The purpose of this session was to familiarise participants with using the devices and how to navigate through virtual space. This session ended when participants felt fully confident and comfortable with the devices to start the formal experiment, which generally lasted approximately three minutes. Afterwards, participants were teleported to the virtual building to start the formal experiment.
- 3. Formal experiment: At the beginning of the formal experiment, participants were initially located in front of room 4.02 in the virtual building. The task information appeared on the screen to instruct participants to begin the first task. Once participants arrived at the task's destination, the next task was depicted. At the beginning of the fourth task, the evacuation alarm sound was automatically triggered. Once the participants reached one of the exits, the formal experiment was terminated and a message popped up showing 'Task complete'.
- 4. Post-experiment questionnaire: Once the participants completed the formal experiment, they were asked to fill in the post-experiment questionnaire in the same experimental room.
- 5. Health check: After filling the questionnaire, the experimenter checked with participants whether they felt any discomfort. Participants were only allowed to leave if they had a normal state of health.

#### 3.5. Data collection and analysis

Two types of data were collected during the experiment, namely the behavioural data pertaining to the participant's movement in the VR environment and questionnaire data pertaining to user experience with the VR system. In order to identify significant differences regarding pedestrian behaviour data and questionnaire data between the two groups, a two-step statistical procedure was applied. First, the Shapiro–Wilk test was conducted to test distribution normality, which is an appropriate normality test for sample sizes that are below 50 [71]. Second, if the data was found to be normally distributed, the independent *t*-test was carried out to determine whether the differences were significant. Otherwise, the nonparametric Mann-Whitney U test was used. For all statistical analyses, the significance level was set at 0.05. Moreover, the effect size was also calculated.

#### 3.5.1. Pedestrian behaviour data

Each participant's behaviour in the virtual environment was recorded. Participant's positions, head rotations (i.e., yaw, pitch, and roll), gaze points, together with timestamp were recorded at a frequency of 10 Hz within UE4. According to the identified wayfinding behaviour metrics in literature (Section 2.1), these data were translated into three



Fig. 5. The layout of the virtual environment, A1-E2 are exits.

types of metrics regarding participant's wayfinding behaviour, namely (1) participant's decision making (e.g., wayfinding strategy, route choice, and exit choice), (2) observation behaviour (e.g., head rotation and gaze point), and (3) wayfinding task performance (i.e., travel time, travel distance, and travel speed). Based on the definition of these metrics in literature, the explanation of each adopted metric for data analysis is listed below:

#### 1. Participant's decision making

- (1) The wayfinding strategy: The usage of the wayfinding strategy of each participant during each task is analysed. Here, we take the start position, room 4.02 as the well-known parts of the building to distinguish between central point strategy and floor strategy, namely to identify whether participants pass room 4.02 during the current task and the precious task. For instance, if one participant passes room 4.02 during task 2 and accordingly passes 4.02 again during task 3, we record the central point strategy; if one goes directly to the fourth floor and does not pass room 4.02 again during task 3, we record the floor strategy for this participant. Fig. 6 illustrates the wayfinding strategies that were adopted by one participant during the experiment. This participant used the central point strategy during task 1 (orange trajectory), the direction strategy during task 2 (green trajectory), and the floor strategy during task 3 (blue trajectory) and task 4 (red trajectory).
- (2) The usage of paths: A path is defined as the smallest section connected by two decision points that located along the two big parallel corridors. The distribution of used paths is analysed.
- (3) The usage of decision points: Since in the current experiment participants face the same number of decision points, we analyse the distribution of decision points that each participant used, namely the decision points where a participant decides to cross from one side to another side of the building.
- (4) The usage of staircases: The distribution of staircases used by participants is analysed. Fig. 7 shows the distribution of one individual's usage of path, decision point, and staircase during task 2.

#### 2. Observation behaviour

In order to better understand participant's observation behaviour during wayfinding tasks, both head rotation data and gaze point data is analysed. For the head rotation analysis, in order to limit noise caused by participants who shake their heads while walking [24], we only focus on head rotation data along the Yaw axis. The Yaw movement is the head rotation on the horizontal plane between  $-180^{\circ}$  and  $180^{\circ}$  (i.e., rotate the head left/right). Participants' average head rotation change  $\overline{Y}$  during each task is calculated by Formula 1 and 2:

$$Y(t) = min(360 - |Y_{t+dt} - Y_t|, |Y_{t+dt} - Y_t|.)$$
(1)

$$\overline{Y} = \frac{\sum_{1}^{T} Y}{T}$$
(2)



Fig. 6. An illustration of different wayfinding strategies adopted by one participant during the experiment.



Fig. 7. Distribution of one individual's usage of path, decision point and staircase during task 2.

where Y(t) is the instantaneous rotation change,  $Y_t$  is the current Yaw coordinate of the participant at *t* timestep and *dt* is the timestep interval, *T* is the travel time of the task.

The point of interest in the virtual environment is determined using a gaze point analysis. Gaze points identify the locations where participants rotate their head and the direction of the head hits the geometry in the environment. Please note, in the current paper, we calculated gaze points based on participants' head directions. We assume that participants always look straight ahead and when participants look at certain objects in the virtual building, the direction of the head hits the geometry in the environment and counts as a gaze point. The density of the dots indicates the number of gaze points, the longer participants looked towards that area and the slower the walking speed of participants passing that area. More sparse distribution of the dots indicates that fewer gaze points were created, which means that participants paid less attention to that area and passed by quickly.

#### 3. Wayfinding task performance

Wayfinding task performance is measured in terms of travel time, travel distance, and average speed of each task in both groups.

Travel time is defined as the time participants spent from the starting location to the destination location during each task. It is one of the important criteria to measure the performance of pedestrian wayfinding tasks [29].

Travel distance is defined as the total distance that participants travelled from the starting location to the destination location during each task. The travel distance is the sum of travel distance on the corridor  $D_c$  and travel distance on the staircase  $D_s$ , which are calculated using Formula 3 and 4:

$$D_{c} = \sum_{1}^{I_{c}} \sqrt{\left(x_{t+dt} - x_{t}\right)^{2} + \left(y_{t+dt} - y_{t}\right)^{2}}$$
(3)

$$D_s = \sum_{1}^{T_s} \sqrt{(x_{t+dt} - x_t)^2 + (y_{t+dt} - y_t)^2 + (z_{t+dt} - z_t)^2}$$
(4)

where  $x_t$ ,  $y_b$ ,  $z_t$ , are the x-coordinate, y-coordinate, and z-coordinate of the participant at *t* timestep, *dt* is the data recording timestep (0.1 s).  $T_c$  is the travel time on the horizontal plane and  $T_s$  is the travel time on the vertical plane.

The average speed per task is computed for each participant by dividing the total travel distance by the total travel time spent during each task.

#### 3.5.2. User experience data

The personal features and experiences of each participant regarding the virtual experiment were collected via the questionnaire. The questionnaire contained five sections: (1) participant's information, which included their socio-demographic information and their experience with VR, computer gaming and the experimental building in reallife, (2) the face validity questionnaire, which assessed whether a simulator measures what it is intended to measure [107], (3) the Simulator Sickness Questionnaire [108], which determined if participant's experience sickness throughout the experiment, (4) the System Usability Scale [109], which assessed the usability of the applied VR systems, and (5) the Presence Questionnaire [110], which measured participant's sense of presence in the virtual environment. This comprehensive questionnaire was used in order to ensure that the authors are able to study and compare user experience in the virtual environment in great detail.

#### 3.6. Participant recruitment & characterisation

A priori power analysis was conducted using G\*Power 3.1 in order to estimate the required sample size [111]. The result indicated that a total sample of 68 participants would be needed to detect large effects (d = 0.80) with 90% power using an independent samples *t*-test between means (two-group). A total sample of 43 participants would be needed to detect large effects (w = 0.50) with 90% power using the Chi-square test. Effect size describes the magnitude of differences found between two groups and larger differences lead to more powerful tests [112]. In this study, the effect size is chosen based on Cohen's definition of large effect size [113] and similar studies [114–115] that adopted effect sizes of 0.7 and 0.69, respectively.

Participants were recruited by means of advertisements at the Delft University of Technology (i.e., e-mails, websites, flyers, posters, social media, and in-classroom promotions). In this stage, potential participants were told that the purpose of the experiment was to investigate the usage of VR to study pedestrian behaviour. In order to not bias participants' behaviour, no information was provided regarding the actual VR experiment.

The VR experiments were carried out from 27th November 2019 to 18th December 2019. In this study, a total sample of 72 participants joined the experiment, 38 participants took part in the HMD group and 34 took part in the Desktop group. All participants volunteered to take part in the experiment and did not receive compensation for their participation. All participants had normal or corrected-to-normal visions and normal hearing capabilities. Two participants from the HMD group asked to take a break during the experiment and did not finish the whole experiment, so they were excluded from further analysis.

The participants were between 22 and 64 years old (M = 27.85, SD = 6.83) in the Desktop group, and between 17 and 41 years old (M = 28.66, SD = 6.00) in the HMD group. The Mann-Whitney U test showed that there was no significant difference in age between the two groups (U = 486.5.5, p = 0.07, d = 0.13). Table 1 presents a summary of the characteristics of the participants in two groups. All participants in both groups had a certain familiarity with the faculty building featuring in the

#### Table 1

#### The descriptive information of participants.

Descriptive information	Category	HMD	Desktop
Gender	Male	17	24
		(47.22%)	(70.59%)
	Female	19	10
		(52.78%)	(29.41%)
Familiarity with the faculty	Not at all familiar	0 (0.00%)	0 (0.00%)
building	A-little familiar	1 (2.78%)	5 (14.71%)
	Moderately familiar	5 (13.88%)	4 (11.76%)
	Quite-a-bit familiar	9 (25.00%)	7 (20.59%)
	Very familiar	21	18
	•	(58.34%)	(52.94%)
Highest education level	High school or equivalent	5 (13.88%)	0 (0.00%)
	Bachelor's degree or	6 (16.67%)	10
	equivalent		(29.41%)
	Master's degree or	19	21
	equivalent	(52.78%)	(61.77%)
	Doctoral degree or	6 (16.67%)	3 (8.82%)
	equivalent		
Previous experience with	Never	11	7 (20.59%)
VR		(30.55%)	
	Seldom	18	15
		(50.00%)	(44.12%)
	Sometimes	6 (16.67%)	9 (26.47%)
	Often	1 (2.78%)	0 (0.00%)
	Very often	0 (0.00%)	3 (8.82%)
Familiarity with any	Not at all familiar	6 (16.67%)	3 (8.82%)
computer gaming	A-little familiar	6 (16.67%)	6 (17.64%)
	Moderately familiar	8 (22.22%)	5 (14.71%)
	Quite-a-bit familiar	7 (19.44%)	7 (20.59%)
	Very familiar	9 (25.00%)	13
			(38.24%)

VR experiment. Most of the participants received a bachelor's degree or higher level of education. More than half of the participants had never or seldom tried VR before; 80.55% in the HMD group and 64.71% in the Desktop group. The familiarity with computer gaming experience was relatively high (i.e., between moderately familiar and very familiar) in the HMD group (66.66%) and the Desktop group (73.54%). Original questions related to 'Familiarity with the faculty building', 'Highest education level', 'Previous experience with VR', and 'Familiarity with any computer gaming' can be found in Appendix A. Chi-square tests showed that there were no significant differences found between two groups regarding gender ( $X^2(1, N = 70) = 0.71, p = 0.399, phi = 0.24$ ), familiarity with the faculty building ( $X^2(3, N = 70) = 3.20, p = 0.361, v$ = 0.21), the highest level of education ( $X^2(3, N = 70) = 7.05, p = 0.070,$ v = 0.32), experience with VR ( $X^2(4, N = 70) = 5.71$ , p = 0.222, v =0.29) and familiarity with computer gaming ( $X^2(4, N = 70) = 2.36, p =$ 0.669, v = 0.18). The Chi-square and Mann-Whitney U tests are statistical hypothesis tests that identify whether a statistically significant difference between the participant population of both groups exists. The results indicated that there are no differences in participants' characteristics (i.e., age, gender, familiarity with the faculty building, highest education level, previous experience with VR, and familiarity with any computer gaming) between the two groups. Thus, participants' characteristics do not impact the further comparison of pedestrian wayfinding behaviour and user experience between the two groups.

#### 4. Results

This study examined the difference in participants' wayfinding behaviour and user experience as a result of the adoption of HMD VR and Desktop VR. To this end, pedestrian wayfinding behaviour is first analysed and compared in Section 4.1. Secondly, the user experience of VR is analysed and compared in terms of realism, simulation sickness, feeling of presence, and system usability in Section 4.2.

#### 4.1. Pedestrian wayfinding behaviour

As mentioned above, pedestrian wayfinding behaviour can be evaluated based on three levels of metrics including decision making, observation behaviour, and wayfinding task performance. Using these metrics, this section presents an analysis of pedestrian behavioural data collected during the VR experiment, namely (1) pedestrian route and evacuation exit choice, (2) observation behaviour, and (3) wayfinding task performance.

#### 4.1.1. Route and evacuation exit choice behaviour

In order to better understand pedestrians' route and exit choice during the wayfinding tasks, this section analyses pedestrian route and exit choice during each task, including (1) the wayfinding strategy, (2) the usage of paths, (3) the usage of decision points, and (4) the usage of staircases.

#### 1. Task 1

Fig. 8 shows the aggregated movement trajectories of participants during the first task (room 4.02 - room 4.99), including the usage of paths and decision points. The central point strategy was employed by 24 (66.67%) participants in the HMD group and 24 participants (70.59%) in the Desktop group. These participants first moved along the corridor where their start position was. In the HMD group,12 participants (33.33%) used the direction strategy while 10 participants (29.41%) in the Desktop group used the direction strategy. They first moved in the direction of the target room. There was no significant difference in wayfinding strategies according to the Chi-square test,  $X^2(1, N = 70) = 0.125, p = 0.98, phi = 0.04$ .

Regarding the usage of paths, the Chi-square test showed there was no significant difference in the usage of paths between the Desktop group and the HMD group during task 1,  $X^2(7, N = 287) = 1.62, p = 0.978, v = 0.08$ .

Fisher exact test showed there was no significant difference in the usage of decision points between the Desktop group and the HMD group during task 1 (p = 0.626, v = 0.31). In total, the number of used decision points was 38 in the HMD groups and 34 in the Desktop group.

#### 2. Task 2

Fig. 9 shows the aggregated movement trajectories of participants during the second task (room 4.99 – room 2.01). In the HMD group and the Desktop group, 27 participants (75.00%) and 28 participants (82.35%) employed the floor strategy, respectively. The direction strategy was employed by 9 participants (25.00%) in the HMD group and 6 participants (17.65%) in the Desktop group. There was no significant difference in adopted wayfinding strategy according to the Chisquare test,  $X^2(1, N = 70) = 0.561$ , p = 0.454, phi = 0.09.

Fisher exact test revealed that there were no significant differences in the usage of paths (p = 0.999, v = 0.16), decision points (p = 0.527, v = 0.52), and staircases (p = 0.999, v = 0.17) between the Desktop group and the HMD group during task 2.

#### 3. Task 3

Fig. 10 shows the aggregated movement trajectories of participants during the third task (room 2.01 – room 4.64). In the HMD group, 26 participants (72.22%) used the floor strategy, and 10 participants (27.78%) employed the direction strategy. In the Desktop group, 33 participants (97.06%) employed the floor strategy and 1 participant (2.94%) employed the direction strategy. There was a significant difference in the wayfinding strategy between two groups according to the Chi-square test,  $X^2(1, N = 70) = 8.144, p = 0.004, phi = 0.34$ .

Fig. 10 illustrates the distribution of the usage of paths, decision points, and staircases in both groups. The Fisher exact test revealed that





b. Desktop group

22

23

there was no significant difference in the usage of paths (p = 0.119, v = 0.25) between the Desktop group and the HMD group during task 3. However, there were significant differences in the usage of decision points (p = 0.021, v = 0.92) and staircases (p = 0.002, v = 0.33) between the two groups during task 3. These results indicated that although the usage of paths was similar, the usage of decision points and staircases were significantly different.

31

٦Г

#### 4. Task 4

2.01

Fig. 11 shows the aggregated movement trajectories of participants during the evacuation task (room 4.64 - an exit). Regarding the usage of the wayfinding strategy, all participants chose to go down first, thus the

floor strategy was employed by both groups. The Chi-square test showed there were no significant differences in the usage of paths,  $X^2(2, N = 72) = 0.230$ , p = 0.891, v = 0.06, and the usage of staircases,  $X^2(5, N = 210) = 0.686$ , p = 0.984, v = 0.06.

15

[1]

Even though five main exits were available, only exits C (i.e., C1 and C2) and D (i.e., D1 and D2) were chosen, which are the nearest two exits for the participants. In the HMD group, 9 participants chose exit C1, 9 participants chose exit C2, 12 participants chose exit D1, and 6 participants chose exit D2. In the Desktop group, 8 participants chose exit C1, 11 participants chose exit C2, 8 participants chose exit D1, and 7 participants chose exit D2. There was no significant difference in the exit usage between the two groups using the Chi-square test,  $X^2(3, N = 70) = 1.079$ , p = 0.782, v = 0.12.





Corridor Room Staircase Elevator x Stair usage x Path usage Decision point usage

1

Fig. 10. Visualisation of the participants' movement trajectories and the frequency of staircase, path and decision point usage during task 3.

#### 4.1.2. Observation behaviour

2.01

For task 1, the distributions of the head rotation change  $\overline{Y}$  of both groups were not normally distributed (Shapiro–Wilk test, p < 0.001). Thus, the non-nonparametric Mann-Whitney U test was performed, which showed that there was a significant difference in head rotation change between the two groups (U = 372, p = 0.002, d = 0.44). The average rotation change of HMD group ( $M = 7.38^{\circ}/s$ ,  $SD = 4.00^{\circ}/s$ ) was significantly higher than the Desktop group ( $M = 5.55^{\circ}/s$ ,  $SD = 4.24^{\circ}/s$ ). Fig. 12 shows the aggregated distributions of participants' gaze points during task 1. Here, room numbers (i.e., red dots along with the rooms), fire doors (i.e., red perpendicular in the corridors), and floor plans (i.e., red perpendicular near the floor plans) were the main attractions. Fig. 12, furthermore, illustrates that participants from the HMD group had denser gaze points near room numbers and fire doors, which indicated that participants had more 'looking around' behaviour and paid more attention to room numbers and fire doors in the HMD group than the Desktop group.

1

For task 2, the Shapiro–Wilk test rejected that the head rotation change of both groups followed a normal distribution (p < 0.05). The Mann-Whitney U test showed that there was a significant difference in the head rotation change between the two groups (U = 461.5, p = 0.039, d = 0.17). Participants had significantly higher head rotation change in the HMD group ( $M = 10.71^{\circ}$ /s,  $SD = 2.52^{\circ}$ /s) than the Desktop group ( $M = 10.17^{\circ}$ /s,  $SD = 3.73^{\circ}$ /s). Fig. 13 shows the aggregated gaze distributions of participants during task 2. The result of head rotation change is further supported by the gaze distributions, which illustrated that the AOI was smaller in the Desktop group than the HMD group. Moreover, the density of the resulting gaze points was higher in the HMD group. Thus, fewer room numbers and fire doors were scanned by the participants in the Desktop group.

For task 3, the normal distribution of head rotation change was rejected for both groups (Shapiro–Wilk test, p < 0.05). The average head rotation change is 13.61°/s ( $SD = 3.41^{\circ}$ /s) and 13.05°/s ( $SD = 3.76^{\circ}$ /s) respectively in the HMD group and the Desktop group. The Mann-

Whitney U test showed there was no significant difference in head rotation change between the two groups (U = 538, p = 0.194, d = 0.16). Fig. 14 shows the gaze distributions of participants during task 3. Also in this task, the gaze points of the room number, fire doors and floor plans were major attractions in the environment for both groups.

For the evacuation task (task 4), the normal distribution of head rotation change was rejected for the Desktop group (Shapiro–Wilk test, p < 0.01) but not for the HMD group (Shapiro–Wilk test, p = 0.196). The Mann-Whitney U test showed that there was a significant difference in rotation change between the two groups (U = 304, p < 0.001, d = 0.91). Participants had significantly higher head rotation change in the Desktop group ( $M = 34.26^{\circ}$ /s,  $SD = 7.87^{\circ}$ /s) than the HMD group ( $M = 27.47^{\circ}$ /s,  $SD = 7.06^{\circ}$ /s). Fig. 15 shows the gaze distributions of participants during task 4, which illustrates that exit signs were the major attractions during the wayfinding task (i.e., the red spheres in the corridor next to the staircase). In both groups, all participants chose to go down using the staircase at the right or left side of room 4.64 after the evacuation alarm was triggered. It also showed that participants made a quick decision to go down directly after seeing the exit signs.

#### 4.1.3. Wayfinding task performance

Participants' average travel time for each task are summarised in Table 2. For both groups, the distribution of travel time was not normally distributed (all p < 0.05). Consequently, the nonparametric Mann-Whitney U test was conducted, which showed significant differences in travel time during task 1 (U = 215, p < 0.01, d = 0.97), task 2 (U = 203, p < 0.001, d = 1.23), task 3 (U = 164, p < 0.001, d = 1.11), and evacuation task (U = 316, p < 0.001, d = 0.62). The tests indicated that participants from the HMD group spent significantly more time during each task than the Desktop group.

The average travel distance of participants during each task is displayed in Table 3. The travel distance for both groups was not normally distributed (p < 0.05). Thus, the Mann-Whitney U test was conducted which showed that there were significant differences in travel distance



Fig. 11. Visualisation of the participants' movement trajectories and the frequency of staircase, path, decision point and evacuation exit usage during task 4.



Fig. 12. Distribution of participants' gaze points during task 1.

between the HMD and the Desktop group during task 1 (U = 286, p < 0.01, d = 0.69), task 2 (U = 248, p < 0.01, d = 0.85), task 3 (U = 209, p < 0.01, d = 0.76), and evacuation task (U = 428.5, p = 0.016, d = 0.34). The results indicated that participants in the HMD group travelled significantly longer distances than the Desktop group during each task. Even though the difference in travel distance is 4.70 m on average, this corresponds to a difference in travel time of 16.95 s on average.

The average travel speed of participants during each task is displayed in Table 4. The normal distribution of travel speed was rejected for the Desktop group during all tasks and only rejected for the HMD group during task 1 and task 3 (Shapiro–Wilk test, p < 0.05). The result of Mann-Whitney U tests indicated that there were significant differences in average travel speed between two groups during task 1 (U = 259.5, p < 0.01, d = 0.97), task 2 (U = 220, p < 0.01, d = 0.82), task 3 (U = 177, p < 0.01, d = 1.27), and task 4 (U = 304, p < 0.01, d = 0.80). That is, the participants in the HMD group had significantly slower average speed during each task than the Desktop group.

#### 4.2. User experience

In order to examine whether technological differences influence the user experience of VR, this section analyses the questionnaire data



Fig. 13. Distribution of participants' gaze points during task 2.

collected from the HMD group and the Desktop group, namely the face validity, the simulation sickness, the feeling of presence, and the system

d = 0.34), and 'realism of the virtual furniture' (U = 525, p = 0.107, d = 0.30).

#### 4.2.1. Face validity

usability.

The assessment of face validity included participants' reported answers of four items, namely the realism of the virtual building, the virtual furniture, the movement abilities, and the evacuation alarm sound (see Appendix B). A 5-point Likert scale ranging from 1 (not at all realistic) to 5 (completely realistic) was used by participants to rate the items, which is a typical scale for Likert response [116].

Table 5 shows the descriptive results of face validity for both groups. Overall, the average total score of the HMD group and the Desktop group was above 4 out of 5, which suggested the virtual environment had a high level of realism. Meanwhile, seventy-five percent of the participants graded the total score above 4 or higher, which strengthens the face validity results (see Fig. 16). The total score of the Desktop group was not normally distributed (p = 0.043), and the total score of the HMD group was normally distributed (p = 0.043). Thus, the non-nonparametric Mann-Whitney U test was performed, which showed that there was no significant difference in the average total score of face validity between the two groups (U = 587.5, p = 0.387, d = 0.08).

In addition, the score of all subscales was not normally distributed (all p < 0.001). The Mann-Whitney U test showed that the 'realism of the movement abilities' is significantly different between the two groups (U = 458.5, p = 0.025, d = 0.46). No significant differences existed related to the items 'realism of the evacuation alarm sound' (U = 499.5, p = 0.054, d = 0.45), 'realism of the virtual building' (U = 503.5, p = 0.074,

#### 4.2.2. Simulation sickness

Simulation sickness is generally defined as the discomfort that arises from using simulated environments [117]. In order to investigate the potential for simulation sickness because of the usage of the VR, the Simulator Sickness Questionnaire [108] was used. Sixteen symptoms were rated respectively on a 4-point Likert scale from 0 (None) to 3 (Severe). Scores of these symptoms can be grouped into Nausea (N), Oculomotor (O) and Disorientation (D) subscales, as well as a total symptom score.

Table 6 shows the mean value and standard deviations of SSQ. For both groups, the distributions of the SSQ score were not normally distributed (p < 0.001). The Mann-Whitney U test showed that there was no significant difference in total SSQ score between the two groups (U =559.5, p = 0.269, d = 0.10). Moreover, no significant differences in the subscales of Nausea (U = 591, p = 0.397, d = 0.11), Oculomotor (U =536.0, p = 0.181, d = 0.09) and Disorientation (U = 603.5, p = 0.460, d =0.08) were found between two groups. The boxplots in Fig. 17 showed that the Desktop group had more outliers on the higher score. We did not find a particular reason for the outliers (e.g., age, familiarity with VR or the building) and removing the outliers did not change the statistical results (p > 0.05).

#### 4.2.3. Sense of presence

The Presence Questionnaire (PQ) [110] was used to assess participants' feeling of presence in the virtual environment. It includes four





Corridor Room Staircase Elevator = Floor plan | Fire door

Fig. 14. Distribution of participants' gaze points during task 3.

subscales, namely involvement, sensory fidelity, immersion, and interface quality. The PQ consists of 29 questions and each question was reported from 1 to 7. The total score was counted by summing the reported scores of the 29 items.

Table 7 shows the statistical results of PQ for both groups. In the current study, the mean score of PQ was 146.00 (SD = 13.63) for the HMD group and 148.50 (SD = 17.86) for the Desktop group. The total score of PQ of the HMD group (p = 0.44) and the Desktop group (p = 0.618) was normally distributed. Therefore, the independent *t*-test was performed, which showed there was no significant difference in the total score of PQ between the HMD group and the Desktop group (t = 0.661, p = 0.511, d = 0.16).

All the sub-scale followed a normal distribution (all p > 0.05). Thus, the independent *t*-test was performed, which showed that there were no significant differences in the four subscales between the two groups. The statistics results of the *t*-test for both groups are shown in Table 7.

#### 4.2.4. Usability

The System Usability Scale (SUS) questionnaire [109], which is commonly used as a usability questionnaire, was adopted in the current study to assess the usability of both VR technologies. The questionnaire consists of 10 items with responses on a 5-point Likert scale, from strongly disagree (1) to strongly agree (5). The total score of SUS ranges from 0 to 100.

The average total score of SUS was 83.75 (SD = 11.92) in the Desktop group and 82.01 (SD = 11.10) in the HMD group, which indicated the effective usability of both systems. The total scores of SUS in the HMD group (p = 0.001) and the Desktop group (p < 0.001) were not normally distributed. Accordingly, the Mann-Whitney U test was used to compare the average total score of SUS in both groups, which identified that there was no significant difference between the two groups (U = 511.50, p =0.119, d = 0.15). Table 8 presents the mean scores and standard deviations of ratings for the ten items in the SUS. The first five items were worded positively, and the second five items were worded negatively. The item 'I thought the system was easy to use' received the highest average score in both groups; 'I found the system unnecessarily complex' and 'I needed to learn a lot of things before I could get going with this system' received the lowest score in, respectively the HMD group and the Desktop group. This indicated that both systems were easy and simple to use.

#### 5. Discussion

This paper aims to compare the adoption of different VR technologies for pedestrian wayfinding studies, via investigating the difference in pedestrian wayfinding behaviour and user experience. Wayfinding experiments with two groups of participants were conducted using either HMD VR or Desktop VR. Four hypotheses were formulated, namely there are significant differences in (H1) route and exit choice behaviour (i.e., wayfinding strategies, usage of paths, decision points, staircases, and exits), (H2) observation behaviour (i.e., head rotation and gaze point), (H3) wayfinding task performance (i.e., time, speed, and distance), and (H4) user experience (i.e., realism, presence, simulation sickness, and usability) between the participants that used Desktop VR and the HMD VR. This section discusses the experimental results with respect to pedestrian behaviour and user experience to answer the above-mentioned hypotheses in Sections 5.1 and 5.2, respectively.

#### 5.1. Differences in pedestrian wayfinding behaviour?

This study characterised the pedestrian wayfinding behaviour in the VR environment by means of a selection of metrics, namely route and exit choice behaviour (i.e., wayfinding strategies, usage of paths, decision points, staircases, and exits), observation behaviour (i.e., head rotation and gaze point) and wayfinding task performance (i.e., time, speed, and distance). Underneath, the findings pertaining to each metric are discussed and compared to the literature.

#### 5.1.1. Difference in pedestrian route and exit choice behaviour This study found a limited significant difference in terms of route and



a. HMD group



Corridor Room Staircase Elevator Floor plan Fire door

Fig.	15.	Distribution	of	participants'	gaze	points	during	task ·	4.
				P	0	P	0		

Table 2	
---------	--

Means and standard deviations of travel time (s) in each task.

Task number	HMD group Mean (SD)	Desktop group Mean (SD)	
Task 1	160.79 (20.19)	144.82 (11.12)	
Task 2	201.30 (18.30)	179.09 (17.86)	
Task 3	140.14 (24.02)	118.59 (12.59)	
Task 4	66.67 (11.37)	58.59 (14.50)	

#### Table 3

Means and standard deviations of travel distance (m) in each task.

Task number	HMD group Mean (SD)	Desktop group Mean (SD)
Task 1	188.68 (4.85)	185.27 (5.10)
Task 2	237.82 (5.52)	232.34 (7.28)
Task 3	163.13 (9.57)	156.05 (8.94)
Task 4	73.34 (8.77)	70.52 (7.80)

exit choice behaviour. Only significant differences pertaining to the detailed behaviour (i.e., wayfinding strategy, usage of staircase, and decisions points during task 3) were recorded. Therefore, hypothesis H1: there is a significant difference in route and exit choice behaviour (i.e., wayfinding strategy, paths, decision points, staircases, and evacuation exit choice) between the participants that adopted Desktop VR and HMD VR, was only partially confirmed. In particular, the usage of wayfinding strategies, decision points, and staircases were significantly different in

Table -	4
---------	---

Means and standard deviations of travel speed (m/s) in each task.

Task number	HMD group Mean (SD)	Desktop group Mean (SD)	
Task 1	1.19 (0.11)	1.28 (0.07)	
Task 2	1.19 (0.09)	1.32 (0.13)	
Task 3	1.18 (0.13)	1.32 (0.08)	
Task 4	1.12 (0.15)	1.24 (0.16)	

#### Table 5

The mean value and standard deviations of the face validity questionnaire.

Item	H <b>MD group</b> Mean (SD)	<b>Desktop group</b> Mean (SD)
Total score of face validity	4.04 (0.36)	4.07 (0.42)
Realism of the movement abilities	3.17 (0.65)	3.50 (0.79)
Realism of the evacuation alarm sound	4.75 (0.44)	4.50 (0.66)
Realism of the virtual building	4.08 (0.60)	4.29 (0.63)
Realism of the virtual furniture	4.17 (0.56)	4.00 (0.55)

the case where the destination was not clear-cut.

Overall, the frequency of adopting a certain wayfinding strategy was found to be similar during task 1, task 2, and task 4. This study illustrates that floor strategy was employed most in wayfinding tasks involved floor changes. Moreover, the frequency of using the floor strategy increased with the task number. This finding can be explained by literature (e.g., [56,118]), which suggests that when a destination is unclear, the



Fig. 16. Comparison of the boxplots of the face validity questionnaire for both groups.

#### Table 6

The mean value and standard deviations of SSQ.

Item	HMD group Mean (SD)	<b>Desktop group</b> Mean (SD)
Total score	15.06 (15.19)	17.27 (26.45)
Nausea	9.80 (14.69)	11.78 (20.07)
Oculomotor	13.69 (12.13)	15.38 (22.35)
Disorientation	16.63 (20.73)	18.83 (32.85)

proportion of participants who use the floor strategy increases. In this study, the destinations of task 1 and task 2 (i.e., the destinations were at the end of the corridor) were clearer, more regular and easier to understand than task 3 (room 4.64). Thus, the unclarity of task 3 might have resulted in a higher proportion of participants employing the floor strategy during task 3. Another explanation is that staircases were very visible and easily accessible in the current experimental environment, thus inviting participants to move to the destination floor at their earliest convenience. A significant difference was only found during task 3 regarding the wayfinding strategy. There are no published studies comparing pedestrian wayfinding strategies using different VR technologies. Some literature suggests that users in an immersive virtual environment were more likely to move directly between junctions [72], namely they like to use the 'straightest' route, which explained why participants from the HMD group were more likely to employ the direction strategy.

Considering the usage of paths, participant's behaviour was found to be overall similar between both groups. That is, both groups had a similar distribution of used paths during all tasks. Besides that, this study shows that participants preferred to use the wider and longer corridors for all tasks. In particular, the pedestrians refrained from using the smaller corridors that connected the two parallel main corridors and predominantly used the larger open areas to cross between the two main corridors. This is in accordance with studies of [64–67] that found participants preferred paths that are wider and with longer lines of sight.

The usage of decision points and staircase were overall similar for task 1, task 2, and task 4 between both groups. A significant difference was only found for task 3 regarding the usage of decision points and staircases. This is because more participants in the HMD group adopted the direction strategy than the Desktop group, which also leads to the different usage of decision points and staircases. Moreover, participants who used Desktop VR have a fixated view and a vantage point to observe the virtual building from the real-life environment, which means that they have a more accurate estimation of the movement direction [99].

Regarding the usage of exits during the evacuation, both groups



Fig. 17. Comparison of the boxplots of the SSQ questionnaire for both groups.

Table 7Subscales of PQ: Means and standard deviations (range from 1 to 7).

Item	HMD group Mean (SD)	Desktop group Mean (SD)	t-value, p-value, d
Involvement	4.81 (0.62)	5.08 (0.83)	1.531, 0.130, 0.37
Sensory fidelity	4.91 (0.87)	4.89 (0.82)	-0.123, 0.903, 0.02
Immersion	5.78 (0.50)	5.56 (0.68)	-1.625, 0.109, 0.37
Interface quality <sup>a</sup>	4.17 (0.97)	4.59 (1.04)	1.747, 0.085, 0.42

<sup>a</sup> Reversed items

showed similar choices. Meanwhile, the results show the usage of the exits was asymmetrical. That is, only the nearest four exits were used among ten available exits. This behaviour is in line with other studies that look at exit usage [21,119–121]. Meanwhile, in both groups, participants only chose the closet exits. This result is also consistent with recent findings that suggest pedestrians are overall more likely to choose the nearest exits and shortest routes [18,20,70,94,122–124].

#### 5.1.2. Difference in observation behaviour

Regarding head rotation change, significant differences in tasks 1, 2, and 4 were found between two groups. Ultimately, hypothesis H2: there is a significant difference in observation behaviour (i.e., head rotation and gaze point) between the participants that adopted Desktop VR and HMD VR, was only partially confirmed.

Generally, room numbers, floor plans, fire doors, and evacuation exit signs were the major attractions for participants to find their way in the virtual environment. This finding is consistent with literature that suggests people pay more attention to salient landmarks and information that aids navigation [66,125]. More head rotation changes were identified in the HMD group than the Desktop group during tasks 1 and 2. This can be explained by literature, which suggests that viewers use a desktop have a higher degree of expectation regarding the direction in which they are likely to travel [99]. The result is consistent with navigation studies that suggest that participants feel the fully immersive VR setting is more natural and intuitive to look around [15,82,126]. Participants in the Desktop group needed to press the mouse to change their view to rotate and their physical view is still fixated on the computer screen in front of them [44], while participants in the HMD group required fewer efforts and felt more natural to rotate (i.e., simply move their head).

Yet, contrary to our expectation, during the evacuation task, participants from the Desktop group had significantly more head rotation changes than the HMD group. This can be explained by two reasons. Firstly, it might be that participants of the HMD group did already observe the environment more during previous tasks, thus less

### Table 8

Mean scores and standard deviations of the SUS questionnaire.

Item	HMD group Mean (SD)	<b>Desktop group</b> Mean (SD)
I thought the system was easy to use	4.39 (0.80)	4.65 (0.65)
I felt very confident using the system	4.33 (0.83)	4.56 (0.75)
I would imagine that most people would learn to use this system very quickly	4.11 (0.92)	4.44 (0.99)
I found the various functions in this system were well integrated	3.92 (0.73)	4.15 (0.61)
I think that I would like to use this system frequently	3.39 (1.15)	3.13 (1.09)
I think that I would need the support of a technical person to be able to use this system	1.56 (0.81)	1.44 (0.99)
I thought there was too much inconsistency in this system	1.56 (0.77)	1.38 (0.60)
I found the system very cumbersome to use	1.53 (0.81)	1.76 (1.10)
I needed to learn a lot of things before I could get going with this system	1.36 (0.80)	1.26 (0.62)
I found the system unnecessarily complex	1.33 (0.53)	1.62 (1.10)

'observing' was required in the follow-up tasks due to the learning effect. Secondly, with increasing task complexity (i.e., find an exit during an emergency), the participants from the Desktop group increased their levels of head rotation to acquire the necessary information from the environment in order to evacuate and find the location of exits. Meanwhile, the results show that when tasks get more complex (i.e., tasks 3 and 4), it is generally more intuitive and natural for participants to look around.

#### 5.1.3. Difference in wayfinding task performance

With respect to the task performance, the results revealed that significant differences existed in participants' travel time, travel distance, and average travel speed during all tasks. Ultimately, hypothesis H 3: there is a significant difference in wayfinding task performance (i.e., time, speed, and distance) between the participants that adopted Desktop VR and HMD VR, was confirmed.

The results indicate that participants had more efficient task performance in the Desktop group compared to the HMD group. This finding is consistent with studies that reported better navigation efficiency in a Desktop VR than an immersive VR [15,44,127]. It problematizes the simple assumption that more immersive VR leads to better performance. This might be explained by three reasons. First, participants in the HMD group used more decision points compared to the Desktop group. As Suzer et al. [29] stated, if a traveller chooses a longer route, the usage of decision points increases, which also leads to more difficulties in the wayfinding task. Therefore, the increase of decision points can cause an increase in the travel distance and travel time. Second, according to the result of head rotation change and gaze points, participants in the HMD group have more 'observing behaviour' (i.e., higher head rotation change, higher density of gaze points, and bigger AOI) than those in the Desktop group, which might cause longer travel time. Moreover, the significant difference in travel time is consistent with [14-15], which found participants spend longer time using HMD VR compared to using Desktop VR in wayfinding tasks. Third, the participants from the Desktop group used a standard computer and had more experience with computer gaming, while most of the participants from the HMD group had never or seldom used VR before. Therefore, participants from the Desktop group operated the system more easily, which might cause a shorter travel time.

#### 5.2. Differences in user experience?

This study characterised the user experience of the participants by means of four assessments, namely the face validity, simulation sickness, sense of presence, and usability. Underneath, the results pertaining to each questionnaire are discussed and compared to the literature.

#### 5.2.1. Face validity

Regarding the face validity questionnaire (see Appendix B), participants in both groups reported an average score above 4 (total score: 5), which confirmed the face validity of both VR setups. This was also confirmed by comments from participants, for instance, 'I feel like walking in the faculty'as well as 'I feel the urge to get out of this building and 'I want to be out of this building as quick as possible' for the evacuation task. The realism of movement abilities from the Desktop group was significantly higher than the HMD group. This finding might be caused by the 'step-by-step' movements in the HMD VR, which cause participants to experience less continuous movements at low walking speeds compared to the smoother movements in the Desktop VR. The score of the face validity questionnaire of the current study is similar to other pedestrian-related studies that also used HMD VR (e.g., [117,128]). However, no comparative 'face validity' studies addressing different VR technologies for pedestrian studies have been found in the literature.

#### 5.2.2. Simulation sickness

With respect to the simulation sickness, the average total score of SSQ in both groups was relatively low considering the maximum total score of the SSQ is 236 [129]. Compared to the study of [21,130], participants from the current study had lower SSQ scores although they had longer exposure time in the virtual environment.

Although it is generally assumed that motion sickness increase from Desktop VR to HMD VR [131], that is not the case in the current study. We found that there was no significant difference in SSQ between both groups, and participants reported a higher average score in the Desktop group than the HMD group. This surprising finding can be explained by three reasons from the literature. First, according to sensory conflict theory [132], simulation sickness is a result of conflicts between visual inputs and vestibular inputs. While participants in the Desktop group were moving in the virtual environment, their bodies remained sitting in the real world. Thus, participant's visual system indicated that they were moving, however, their vestibular system told the body it was stationary. In the HMD group, participants continuously rotated their heads in real life to change the direction of movement in the virtual environment, thus fewer sensory conflicts were expected. Second, participants in the HMD group had more active searching behaviour (indicated by the observation behaviour) than the Desktop group. Literature suggests that active participants may experience fewer symptoms [99]. Third, there were more outliers of the SSQ scores of the Desktop group which might cause a, on average, higher SSQ score.

Besides that, in both groups, Disorientation received the highest score, followed by Oculomotor and Nausea. Although the Disorientation subscale is related to vestibular disturbances, such as dizziness and vertigo, high disorientation may be an indicator that participants experienced higher levels of virtual presence [133]. The relatively high disorientation score might result from response lags. The current experiment tasks involved changing floors and some turning movements on the stairs in the virtual building, which are key sources of disorientation about one's heading and position in a building. The relation between disorientation and floor changes was also found in Hölscher et al. [58].

#### 5.2.3. Sense of presence

In terms of the feeling of presence, the results revealed that participants in both groups experienced a similarly high level of presence. Moreover, the PQ scores in the current study are also slightly higher than the studies that also used VR technologies to study pedestrian behaviour (e.g., [71,79,117]). When studying the subscales of PQ, the subscale 'Immersion' received the highest scores in both groups, which identifies that participants felt enveloped by, included in and interacted realistically with the virtual environment. In literature, generally the Desktop VR is categorised as 'non-immersive' VR and the HMD VR is 'immersive VR'. Yet, although the average score of the sub-scale Immersion was slightly higher in the HMD group in the current study, we did not find significant differences between two groups. This finding indicates that both VR technologies provided a similar immersion effect to participants. Our finding is in agreement with the study of [43], which showed that Desktop VR can also provide a good sense of presence to users.

We think there are two potential reasons for the finding that participants from the Desktop group reported similar levels of presence as the HMD group. One explanation is provided by literature, which suggests that when users are more comfortable and less focused on the interaction with VR technology itself (and more with their task), the feeling of presence increases [42]. In our study, the wayfinding tasks force participants' attention away from the interaction with VR. However, in the HMD group, participants needed to wear a headset all the time and proactively rotate their heads to search for information, while participants in the Desktop group sat comfortably in front of a screen, and simply used a mouse and keyboard to move. It means that participants in the HMD group might be less focused on the wayfinding tasks because they were distracted by the headset. A recent discussion by [134] provides another potential explanation, namely presence is related to users' expectations. People normally have a lower expectation of fidelity and 'realism' of non-immersive VR systems than immersive VR systems. Thus, the lower expectations on Desktop VR limit the negative reports on the feeling of presence when using Desktop VR.

#### 5.2.4. Usability

Participants confirmed the usability of the two systems were both at an 'excellent level', according to the interpretation of the SUS score made by Bangor et al. [135], which confirmed that both VR setups had good usability. Meanwhile, the SUS score of both groups is higher than the studies of [21,97,117,136], which are the only four studies that measured SUS regarding the usage of VR in pedestrian behaviour. In both groups, the item 'I thought the system was easy to use' received the highest score, which indicate that both VR setups were easy to use. The slightly higher average score of participants in the Desktop group might be due to their previous experience using a computer, as suggested by literature [97]. That is, participants were more familiar with the display, mouse, and keyboard interface.

To summarise, there were no significant differences in terms of realism, simulation sickness, the feeling of presence, and usability between the two groups. Thus, hypothesis H4: there is a significant difference in user experience (i.e., realism, presence, simulation sickness, and usability) between the participants that adopted Desktop VR and HMD VR, was rejected in this study.

#### 5.3. Implications

Based on the key findings of this study, we highlight several theoretical and practical implications that are both relevant for pedestrian wayfinding research and human–computer interaction research.

#### 5.3.1. Implications for theory

This study provides several theoretical implications for pedestrian wayfinding behaviour study and human–computer interaction. Firstly, this study provides empirical evidence that VR can be used to collect pedestrian wayfinding data in complex and multi-level buildings.

Compared to VR studies featured more simplified environments (e.g., [14-15,39]), the current study shows that it is possible to collect adequate behavioural data in complex environments with high experimental control and let participants experience the virtual world in an immersive and engaging way. Secondly, this study identifies that it is possible to collect detailed behavioural data (i.e., movement trajectory, head movement, and gaze points), personal characteristics (e.g., age, gender, and familiarity with VR), and user experience (i.e., realism, simulation sickness, feeling of presence, and usability) using a VR research tool in combination with questionnaires, which is difficult to achieve under real-world conditions. This provides proof that pedestrian researchers can collect comprehensive data sets featuring multidimensional behaviour data simultaneously and provide researchers with new perspectives to understand pedestrian wayfinding behaviour. Moreover, this study provides exemplars of designing VR experiments regarding the experimental set-up that combines the usage of VR technology and questionnaire to study pedestrian wayfinding behaviour and human-computer interaction. Thirdly, this study shows that when applying VR to study pedestrian behaviour, it is also worthy and important to quantitatively investigate the interaction between people and technologies. Human performance in the virtual environment is influenced by the interaction between the individual and the virtual environment (e.g., presence, usability, realism, and simulation sickens) [92-93]. For previous studies that investigated the interaction between people and VR technologies, some were qualitative or exploratory, or they only considered limited perspectives. The current study combines four different questionnaires and compares these perspectives quantitatively. In contrast to previous VR comparison studies (e.g., [99,131,137]), the realism, simulation sickness, presence, and usability in the current study was overall similar between the HMD and the Desktop group. When applying VR, researchers should take extra caution when making assumptions about the interaction between people and VR technologies in various contexts.

This study identifies significant differences in wayfinding task performance between the HMD and the Desktop group. Participants of the Desktop group navigated more quickly and efficiently during all wayfinding tasks. Meanwhile, pedestrian route and exit choice behaviour (i. e., usage of wayfinding strategy, path, decision points, staircase, and exit) were found to be overall similar during the first two and the evacuation tasks. Our study shows that for 'simpler' wayfinding tasks in multi-level buildings, pedestrian route and exit choice behaviour can be measured effectively using a more simple and less expensive Desktop VR. It indicates that for large-scale virtual environments, the benefits gained from increasing immersion may not be as prevalent as suggested in the literature (e.g., [15,137]). Furthermore, the user experience (i.e., realism, simulation sickness, presence, and usability) was overall similar between the two groups. These findings imply that studies that investigate pedestrian route and exit choice behaviour of 'simple' wayfinding tasks (i.e., requires less spatial understanding) do not need to be limited to using immersive VR. Researchers can choose the best practice between HMD VR and Desktop VR base on their budget, existing equipment, and technical supports.

However, the findings regarding the comparison of the wayfinding strategy and observation behaviour imply that differences can appear, especially when (more complex) searching behaviour is triggered. In particular, there were differences in route choice (i.e., wayfinding strategy, decision point, and staircase) during the task where the location of the destination was not clear-cut. Meanwhile, participants who used HMD VR had more head rotation changes and observation behaviour during the first two tasks, while participants in the Desktop group had higher head rotation changes during the evacuation task. Thus, in cases where the wayfinding task become more complex and searching behaviour are important factors that aid wayfinding, there may still be advantages to use immersive VR. One is advised to carefully consider the differences in behavioural outcomes between both VR technologies when investigating wayfinding behaviour for more 'complex' wayfinding tasks (i.e., requires more spatial understanding). The findings highlight that if one wants to investigate pedestrian wayfinding and observation behaviour in complex environments, a more intuitive and natural VR setting (HMD VR in our case) needs to be ensured in order to allow natural observation of the environment.

#### 5.3.2. Implications for practice

This study provides insights for designing infrastructure and signage to facilitate wayfinding in complex buildings. This study found that room numbers, floor plans, and fire doors were the main attractions for participants to find their way in a multi-level building. This information helps to identify locations where pedestrians search for information and determine what environmental features they look at in order to inform their wayfinding process. These insights could be useful for practitioners who are involved in planning complex buildings to design effective signage in complex and large-scale buildings. Additionally, our study illustrates that floor strategy was employed dominantly in a multi-level building and pedestrians preferred to use the wider corridors over narrow ones. These findings regarding the usage of wayfinding strategies in a multi-level environment provide empirical evidence for professionals to predict and plan the main navigational flow evenly when a complex network of paths and decision points exist.

Moreover, this study provides proof that VR can be used to study pedestrian wayfinding behaviour and collect valid behavioural data in multi-level buildings. The combination of VR, BIM, and digital twin can be used for engineers through their design process to test the interaction between pedestrians and the built environment that is either too complex to test in real life (e.g., evacuation intervention) or hypothetical scenarios that do not exist today (e.g., new building designs) [12]. With the advantages provided by VR, such as the flexibility to simulate a variety of contexts and repeatability to conduct experiments with similar settings, it helps engineers deal with the increasing complexity of the modern engineering systems [138] and evaluate the design, which can be used to improve the design in an early stage of the design process before it is finally implemented. Therefore, provide engineers benefits of shortening development times, lowering construction costs, and construction risks.

#### 6. Conclusion and future work

This study investigated differences in pedestrian wayfinding behaviour in a multi-level building and user experience of the VR technology in order to compare the adoption of HMD VR and Desktop VR for pedestrian wayfinding studies. In particular, pedestrian behavioural data (i.e., pedestrian route and exit choice behaviour, observation behaviour, and wayfinding task performance) and user experience data (i.e., realism, simulation sickness, presence, and usability) were compared between two groups of participants, where one group used the HMD VR and another group used the Desktop VR.

This study provides the first direct comparison between VR technologies regarding the differences in behavioural outcomes of pedestrians in a multi-level building and user experience between the adoption of different VR technologies. It provides the first solid empirical evidence of direct comparison between HMD VR and Desktop VR on the resulting wayfinding behaviour in a multi-level building, which can have an important implication for future investigation of pedestrian wayfinding and evacuation behaviour in complex buildings. The comparison between HMD VR and Desktop VR implies studies that investigate pedestrian route and exit choice behaviour of 'simple' wayfinding tasks (i.e., requires less spatial understanding) do not need to be limited to using immersive VR. However, if one wants to investigate pedestrian wayfinding and observation behaviour in complex environments, a more intuitive and natural VR setting (e.g., HMD VR) is recommended in order to allow natural observation of the environment.

Based on the key findings, we highlight several contributions of the current study. Firstly, this study applies emerging technologies to study

pedestrian wayfinding behaviour. Our findings show that one can study pedestrian wayfinding behaviour in complex environments (even in emergency situations) using VR with high experimental control and still collect comprehensive data (i.e., movement trajectory, head rotation, gaze point, and user experience). This supports researchers and engineers to leverage their understanding of pedestrian behaviour in complex environments with new possibilities and perspectives. Secondly, the current study provides a quantitative and comprehensive comparison of pedestrian wayfinding behaviour and user experience as a result of the adoption of different VR technologies. It offers both theoretical underpinnings of the similarities and differences in pedestrian wayfinding behaviour using different VR technologies and practical suggestions for researchers who are interested in applying VR. Thirdly, this study provides clear Desktop VR and HMD VR use-cases of designing VR experiments regarding the experimental set-up, which combines the usage of VR technology and questionnaire to study pedestrian wayfinding behaviour and user experience. The combination may assist researchers in associating between pedestrian behaviour and human-computer interaction in order to better understand the usage of VR in the pedestrian field and formalize complex engineering knowledge.

Yet, there are several limitations of the current study. Firstly, in order to quantitatively measure the advantage of using HMD VR for complex wayfinding tasks, future studies should directly compare pedestrian behaviour in real and virtual environments. Due to COVID-19, it has until now been impossible to conduct the field experiment with an identical setting as the VR experiment in the faculty building. Secondly, comments made by the participants during the experiment reveal that additional elements can potentially make the current virtual experience more realistic, such as the presence of other pedestrians and the interaction with other people. In the present study, the virtual environment did not include other agents. Thus, it would be interesting to investigate the impact of VR technology on pedestrian wayfinding behaviour, while including the interaction with other pedestrians. Thirdly, the gazing behaviour of participants was only studied qualitatively in this study and was based on head rotations alone. In future studies, incorporation of precise eye-tracking technologies would allow for a more in-depth (quantitative) analysis of gazing behaviour, such as gaze time, gaze quantity, and gaze sequences. This would improve our understanding pertaining to pedestrians' virtual attention in the environment and pedestrians' decision-making process during wayfinding tasks. Moreover, the combination of behavioural data and physiological data may provide additional insights and new perspectives for wayfinding behaviour. For instance, integrating VR technologies with physiological sensing technologies (e.g., heart rate, electroencephalogram sensors, and biosensors) enables researchers to study participant's wayfinding behaviour in combination with their mental and physical states in the virtual environments with stimuli (e.g., light, sound, signals, and text messages). Lastly, although no differences in personal characteristics were found between the two groups, literature does suggest personal characteristics could influence wayfinding behaviour (e.g., [78,79]). Therefore, future work should explore the impact of individual differences (e.g., familiarity with the environment, gender, and age) on pedestrian wayfinding behaviour in a complex multi-level building to gain a better understanding of individuals performing wayfinding tasks.

#### **Declaration of Competing Interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Acknowledgments

We thank the China Scholarship Council for its financial contribution. Meanwhile, this research is also supported by the ALLEGRO project, which is financed by the European Research Council (Grant Agreement No. 669792).

## Appendix A. List of questions related to participant's characteristics

Are you familiar with the building of the Civil engineering and geosciences Faculty?

The highest education level you achieved.

Are you familiar with any computer gaming?

How often do you experience a virtual reality environment (e.g., gaming, training, entertainment)?

#### Appendix B. Face validity questionnaire

Instruction: Please characterize your experience in the virtual environment with a 5-point scale. Please consider the entire scale when making your responses, as the intermediate levels may apply.

1. How realistic is the virtual building?

1 is Not at all realistic, and 5 is Completely realistic

2. How realistic is the virtual furniture (chairs, doors, etc.)?

- 1 is Not at all realistic, and 5 is Completely realistic
- 3. How realistic is the visual experience of the movement abilities?
- 1 is Not at all realistic, and 5 is Completely realistic
- 4. How realistic is the evacuation alarm sound?

1 is Not at all realistic, and 5 is Completely realistic

#### References

- T.J. Shields, K.E. Boyce, A study of evacuation from large retail stores, Fire Saf. J. 35 (2000) 25–49, https://doi.org/10.1016/S0379-7112(00)00013-8.
- [2] K.J. Jeffery, A. Jovalekic, M. Verriotis, R. Hayman, Navigating in a threedimensional world, Behav. Brain Sci. 36 (2013) 523–587, https://doi.org/ 10.1017/S0140525X12002476.
- [3] K. Andree, D. Nilsson, J. Eriksson, Evacuation experiments in a virtual reality high-rise building: exit a and waiting time for evacuation elevators, Fire Mater. (2015) 4B, https://doi.org/10.1002/fam.
- [4] S.F. Kuliga, B. Nelligan, R.C. Dalton, S. Marchette, A.L. Shelton, L. Carlson, C. Hölscher, Exploring individual differences and building complexity in wayfinding: the case of the seattle central library, Environ. Behav. 51 (2019) 622–665, https://doi.org/10.1177/0013916519836149.
- [5] M. Kruminaite, S. Zlatanova, Indoor space subdivision for indoor navigation, in: GIS Proc, ACM Int. Symp. Adv. Geogr. Inf. Syst. (2014) 25–31, https://doi.org/ 10.1145/2676528.2676529.
- [6] E.R. Galea, S.J. Deere, C.G. Hopkin, H. Xie, Evacuation response behaviour of occupants in a large theatre during a live performance, Fire Mater. 41 (2017) 467–492, https://doi.org/10.1002/fam.2424.
- [7] S. Heliövaara, J.M. Kuusinen, T. Rinne, T. Korhonen, H. Ehtamo, Pedestrian behavior and exit selection in evacuation of a corridor - An experimental study, Saf. Sci. 50 (2012) 221–227, https://doi.org/10.1016/j.ssci.2011.08.020.
- [8] M. Imanishi, T. Sano, Route choice and flow rate in theatre evacuation drills: analysis of walking trajectory data-set, Fire Technol. 55 (2019) 569–593, https:// doi.org/10.1007/s10694-018-0783-2.
- [9] M. Kobes, I. Helsloot, B. de Vries, J.G. Post, N. Oberijé, K. Groenewegen, Way finding during fire evacuation; an analysis of unannounced fire drills in a hotel at night, Build. Environ. 45 (2010) 537–548, https://doi.org/10.1016/j. buildenv.2009.07.004.
- [10] D. Nilsson, A. Johansson, Social influence during the initial phase of a fire evacuation-Analysis of evacuation experiments in a cinema theatre, Fire Saf. J. 44 (2009) 71–79, https://doi.org/10.1016/j.firesaf.2008.03.008.
- [11] A. Rahouti, R. Lovreglio, S. Gwynne, P. Jackson, S. Datoussaïd, A. Hunt, Human behaviour during a healthcare facility evacuation drills: investigation of preevacuation and travel phases, Saf. Sci. 129 (2020), 104754, https://doi.org/ 10.1016/j.ssci.2020.104754.
- [12] Y. Feng, D. Duives, W. Daamen, S. Hoogendoorn, Data collection methods for studying pedestrian behaviour: a systematic review, Build. Environ. 187 (2021), 107329, https://doi.org/10.1016/j.buildenv.2020.107329.
- [13] E. Vilar, F. Rebelo, P. Noriega, Indoor human wayfinding performance using vertical and horizontal signage in virtual reality, Hum. Factors Ergon. Manuf. Serv. Ind. 24 (2014) 601–615, https://doi.org/10.1002/hfm.
- [14] T.J.T. Hsieh, Y.H. Kuo, C.K. Niu, Utilizing HMD VR to improve the spatial learning and wayfinding effects in the virtual maze, in: Int. Conf. Human-Computer Interact, Springer International Publishing, 2018, pp. 38–42.
- [15] B.S. Santos, P. Dias, A. Pimentel, J.W. Baggerman, C. Ferreira, S. Silva, J. Madeira, Head-mounted display versus desktop for 3D navigation in virtual reality: a user study, Multimed. Tools Appl. 41 (2009) 161–181, https://doi.org/ 10.1007/s11042-008-0223-2.

- [16] J. Lin, L. Cao, N. Li, Assessing the influence of repeated exposures and mental stress on human wayfinding performance in indoor environments using virtual reality technology, Adv. Eng. Informatics. 39 (2019) 53–61, https://doi.org/ 10.1016/i.aci.2018.11.007.
- [17] C. Tang, W. Wu, C. Lin, Using virtual reality to determine how emergency signs facilitate way-finding, Appl. Ergon. 40 (2009) 722–730, https://doi.org/ 10.1016/j.apergo.2008.06.009.
- [18] Z. Fang, W. Song, J. Zhang, H. Wu, Experiment and modeling of exit-selecting behaviors during a building evacuation, Physica A. 389 (2010) 815–824, https:// doi.org/10.1016/j.physa.2009.10.019.
- [19] K. Fridolf, E. Ronchi, D. Nilsson, H. Frantzich, Movement speed and exit choice in smoke- filled rail tunnels, Fire Saf. J. 59 (2013) 8–21, https://doi.org/10.1016/j. firesaf.2013.03.007.
- [20] K.J. Zhu, Q. Shi, Experimental study on choice behavior of pedestrians during building evacuation, Procedia Eng. 135 (2016) 206–215, https://doi.org/ 10.1016/j.proeng.2016.01.110.
- [21] Y. Feng, D.C. Duives, S.P. Hoogendoorn, Using virtual reality to study pedestrian exit choice behaviour during evacuations, Saf. Sci. 137 (2021), 105158, https:// doi.org/10.1016/j.ssci.2021.105158.
- [22] J. Lin, R. Zhu, N. Li, B. Becerik-Gerber, Do people follow the crowd in building emergency evacuation? A cross-cultural immersive virtual reality-based study, Adv. Eng. Informatics. 43 (2020), 101040, https://doi.org/10.1016/j. aei.2020.101040.
- [23] G. Cosma, E. Ronchi, D. Nilsson, Way-finding lighting systems for rail tunnel evacuation: a virtual reality experiment with Oculus Rift®, J. Transp. Saf. Secur. 8 (2016) 101–117, https://doi.org/10.1080/19439962.2015.1046621.
- [24] M. Zhang, J. Ke, L. Tong, X. Luo, Investigating the influence of route turning angle on compliance behaviors and evacuation performance in a virtual-realitybased experiment, Adv. Eng. Informatics. 48 (2021), 101259, https://doi.org/ 10.1016/j.aei.2021.101259.
- [25] A. Birenboim, P.B.N. Bloom, H. Levit, I. Omer, The study of walking, walkability and wellbeing in immersive virtual environments, Int. J. Environ. Res. Public Health. 18 (2021) 1–18, https://doi.org/10.3390/ijerph18020364.
- [26] L. Cao, J. Lin, N. Li, A virtual reality based study of indoor fi re evacuation after active or passive spatial exploration, Comput. Human Behav. 90 (2019) 37–45, https://doi.org/10.1016/j.chb.2018.08.041.
- [27] E. Duarte, F. Rebelo, J. Teles, M.S. Wogalter, Behavioral compliance for dynamic versus static signs in an immersive virtual environment, Appl. Ergon. 45 (2014) 1367–1375, https://doi.org/10.1016/j.apergo.2013.10.004.
- [28] M. Kinateder, W.H. Warren, K.B. Schloss, What color are emergency exit signs? Egress behavior differs from verbal report, Appl. Ergon. 75 (2019) 155–160, https://doi.org/10.1016/j.apergo.2018.08.010.
- [29] O.K. Suzer, N. Olgunturk, D. Guvenc, The effects of correlated colour temperature on wayfinding: a study in a virtual airport environment, Displays. 51 (2018) 9–19, https://doi.org/10.1016/j.displa.2018.01.003.
- [30] A. Tucker, K.L. Marsh, T. Gifford, X. Lu, P.B. Luh, R.S. Astur, The effects of information and hazard on evacuee behavior in virtual reality, Fire Saf. J. 99 (2018) 1–11, https://doi.org/10.1016/j.firesaf.2018.04.011.
- [31] R. Zhu, J. Lin, B. Becerik-Gerber, N. Li, Human-building-emergency interactions and their impact on emergency response performance: a review of the state of the art, Saf. Sci. 127 (2020), 104691, https://doi.org/10.1016/j.ssci.2020.104691.
- [32] N.W.F. Bode, A.U. Kemloh Wagoum, E.A. Codling, Human responses to multiple sources of directional information in virtual crowd evacuations, J. R. Soc. Interface. 11 (2014), https://doi.org/10.1098/rsif.2013.0904.
- [33] N.W.F. Bode, A.U. Kemloh Wagoum, E.A. Codling, Information use by humans during dynamic route choice in virtual crowd evacuations, R. Soc. Open Sci. 2 (2015), https://doi.org/10.1098/rsos.140410.
- [34] J.F. Silva, J.E. Almeida, R.J. Rossetti, A.L. Coelho, A serious game for EVAcuation training, Conf. Serious Games Appl. Heal. (2013) 1–6, https://doi.org/10.1109/ SeGAH.2013.6665302.
- [35] M. Moussaïd, M. Kapadia, T. Thrash, R.W. Sumner, M. Gross, D. Helbing, C. Hölscher, Crowd behaviour during high-stress evacuations in an immersive virtual environment, J. R. Soc. Interface. 13 (2016) 20160414, https://doi.org/ 10.1098/rsif.2016.0414.
- [36] M. Kinateder, E. Ronchi, D. Gromer, M. Müller, M. Jost, M. Nehfischer, A. Mühlberger, P. Pauli, Social influence on route choice in a virtual reality tunnel fire, Transp. Res. Part F Traffic Psychol. Behav. 26 (2014) 116–125, https://doi. org/10.1016/j.trf.2014.06.003.
- [37] E. Ronchi, D. Mayorga, R. Lovreglio, J. Wahlqvist, D. Nilsson, Mobile-powered head-mounted displays versus cave automatic virtual environment experiments for evacuation research, Comput. Animat. Virtual Worlds. 30 (2019), e1873, https://doi.org/10.1002/cav.1873.
- [38] D. Bauer, V. Settgast, H. Schrom-Feiertag, A. Millonig, Making the usage of guidance systems in pedestrian infrastructures measurable using the virtual environment DAVE, Transp. Res. Part F Traffic Psychol. Behav. 59 (2018) 298–317, https://doi.org/10.1016/j.trf.2018.09.012.
- [39] R.A. Ruddle, P. Péruch, Effects of proprioceptive feedback and environmental characteristics on spatial learning in virtual environments, Int. J. Hum. Comput. Stud. 60 (2004) 299–326, https://doi.org/10.1016/j.ijhcs.2003.10.001.
- [40] T.S. Mujber, T. Szecsi, M.S.J. Hashmi, Virtual reality applications in manufacturing process simulation, J. Mater. Process. Technol. 155 (2004) 1834–1838, https://doi.org/10.1016/j.jmatprotec.2004.04.401.
- [41] J. Seibert, D.M. Shafer, Control mapping in virtual reality: effects on spatial presence and controller naturalness, Virtual Real. 22 (2018) 79–88, https://doi. org/10.1007/s10055-017-0316-1.

- [42] J.L. Soler-Domínguez, C. de Juan, M. Contero, M. Alcañiz, I walk, therefore I am: a multidimensional study on the influence of the locomotion method upon presence in virtual reality, J. Comput. Des. Eng. 7 (2020) 1–14, https://doi.org/ 10.1093/jcde/qwaa040.
- [43] A.L.E. Lee, K.W. Wong, C.C. Fung, How does desktop virtual reality enhance learning outcomes? A structural equation modeling approach, Comput. Educ. 55 (2010) 1424–1442, https://doi.org/10.1016/j.compedu.2010.06.006.
- [44] H. Li, N.A. Giudice, The effects of immersion and body-based rotation on learning multi-level indoor virtual environments, in: Proc. 5th ACM SIGSPATIAL Int, Work. Indoor Spat. Awareness, ISA 2013 (2013) 8–15, https://doi.org/10.1145/ 2533810.2533811.
- [45] M. Raubal, M.J. Egenhofer, Comparing the complexity of wayfinding tasks in built environments, Environ. Plan. B Plan. Des. 25 (1998) 895–913.
- [46] K. Lynch, The Image of the City, MIT Press, 1960.
- [47] S. Jamshidi, D. Pati, A narrative review of theories of wayfinding within the interior environment, Heal. Environ. Res. Des. J. 14 (2020) 290–303, https://doi. org/10.1177/1937586720932276.
- [48] R. Passini. Wayfinding in architecture, 2nd ed., Van Nostrand Reinhold Company, New York, 1992.
- [49] P. Arthur, R. Passini, Wayfinding: People, Signs, and Architecture, McGraw-Hill Ryerson, Toronto, 1992.
- [50] R. Passini, Spatial representations, a wayfinding perspective, J. Environ. Psychol. 4 (1984) 153–164, https://doi.org/10.1016/S0272-4944(84)80031-6.
- [51] A. Natapov, D. Fisher-Gewirtzman, Visibility of urban activities and pedestrian routes: an experiment in a virtual environment, Comput. Environ. Urban Syst. 58 (2016) 60–70, https://doi.org/10.1016/j.compenvurbsys.2016.03.007.
- [52] M. Raubal, S. Winter, Enriching wayfinding instructions with local landmarks, Lect. Notes Comput. Sci. (Including Subser. Lect. Notes Artif. Intell. Lect. Notes Bioinformatics) 2478 (2002) 243–259, https://doi.org/10.1007/3-540-45799-2\_ 17.
- [53] L.B. Zomer, F. Schneider, D. Ton, S. Hoogendoorn-Lanser, D. Duives, O. Cats, S. Hoogendoorn, Determinants of urban wayfinding styles, Travel Behav. Soc. 17 (2019) 72–85, https://doi.org/10.1016/j.tbs.2019.07.002.
- [54] P. Kiefer, I. Giannopoulos, M. Raubal, Where am i? Investigating map matching during self-localization with mobile eye tracking in an urban environment, Trans. GIS. 18 (2014) 660–686, https://doi.org/10.1111/tgis.12067.
- [55] M. Raubal, M. Worboys, A formal model of the process of wayfinding in built environments, Int. Conf. Spat. Inf. Theory (1999) 381–399, https://doi.org/ 10.1007/3-540-48384-5\_25.
- [56] H. Li, T. Thrash, C. Hölscher, V.R. Schinazi, The effect of crowdedness on human wayfinding and locomotion in a multi-level virtual shopping mall, J. Environ. Psychol. 65 (2019), https://doi.org/10.1016/j.jenvp.2019.101320.
- [57] L.A. Carlson, C. Hölscher, T.F. Shipley, R. Conory Dalton, Getting lost in buildings, Curr. Dir. Psychol. Sci. 19 (2010) 284–289, https://doi.org/10.1177/ 0963721410383243.
- [58] C. Hölscher, T. Meilinger, G. Vrachliotis, M. Brösamle, M. Knauff, Finding the way inside: linking architectural design analysis and cognitive processes, Spat. Cogn. IV. Reason. Action, Interact. 3343 (2005) 1–23.
- [59] Soeda M., Kushiyama N., Ryuzo O., Wayfinding in cases with vertical motion, in: Proc. MERA 97, 1997, pp. 559–564.
- [60] R.A. Ruddle, S. Lessels, Three levels of metric for evaluating wayfinding, Presence Teleoperators Virtual Environ. (2006) 637–654, https://doi.org/10.1162/ pres.15.6.637.
- [61] C. Hölscher, T. Meilinger, G. Vrachliotis, M. Brösamle, M. Knauff, Up the down staircase: wayfinding strategies in multi-level buildings, J. Environ. Psychol. 26 (2007) 284–299, https://doi.org/10.1016/j.jenvp.2006.09.002.
- [62] G. Best, Direction finding in large buildings, Archit. Psychol. (1970) 72–91, https://doi.org/10.1016/0003-6870(72)90130-5.
- [63] S. Jamshidi, M. Ensafi, D. Pati, Wayfinding in interior environments: an integrative review, Front. Psychol. 11 (2020) 1–24, https://doi.org/10.3389/ fpsyg.2020.549628.
- [64] E. Vilar, F. Rebelo, P. Noriega, J. Teles, C. Mayhorn, The influence of environmental features on route selection in an emergency situation, Appl. Ergon. 44 (2013) 618–627, https://doi.org/10.1016/j.apergo.2012.12.002.
- [65] E. Vilar, F. Rebelo, P. Noriega, E. Duarte, C.B. Mayhorn, Effects of competing environmental variables and signage on route-choices in simulated everyday and emergency wayfinding situations, Ergonomics. 57 (2014) 511–524, https://doi. org/10.1080/00140139.2014.895054.
- [66] J.M. Wiener, C. Hölscher, S. Büchner, L. Konieczny, Gaze behaviour during space perception and spatial decision making, Psychol. Res. 76 (2012) 713–729, https://doi.org/10.1007/s00426-011-0397-5.
- [67] J. Frankenstein, S. Brüssow, F. Ruzzoli, C. Hölscher, The language of landmarks: the role of background knowledge in indoor wayfinding, Cogn. Process. 13 (2012) 165–170, https://doi.org/10.1007/s10339-012-0482-8.
- [68] J. Weisman, Evaluating architectural legibility: way-finding in the built environment, Environ. Behav. 13 (1981) 189–204, https://doi.org/10.1177/ 0013916581132004.
- [69] I. Omer, R. Goldblatt, The implications of inter-visibility between landmarks on wayfinding performance: an investigation using a virtual urban environment 31 (2007) 520–534, https://doi.org/10.1016/j.compenvurbsys.2007.08.004.
- [70] R.Y. Guo, H.J. Huang, S.C. Wong, Route choice in pedestrian evacuation under conditions of good and zero visibility: experimental and simulation results, Transp. Res. Part B Methodol. 46 (2012) 669–686, https://doi.org/10.1016/j. trb.2012.01.002.
- [71] R. Zhu, J. Lin, B. Becerik-gerber, N. Li, Influence of architectural visual access on emergency wayfinding: a cross-cultural study in China, United Kingdom and

United States, Fire Saf. J. 113 (2020), 102963, https://doi.org/10.1016/j. firesaf.2020.102963.

- [72] R. Conroy, Spatial Navigation in Immersive Virtual, Environments (2001).
- [73] Y. Feng, D.C. Duives, S.P. Hoogendoorn, The impact of guidance information on exit choice behavior during an evacuation—A VR study, in: I. Zuriguel, A. Garcimartin, R. Cruz (Eds.), Traffic Granul. Flow 2019, Springer International Publishing, Cham, 2020, pp. 69–75, https://doi.org/10.1007/978-3-030-55973-
- [74] C. Hölscher, S.J. Büchner, T. Meilinger, G. Strube, Map Use and wayfinding strategies in a multi-building ensemble, Int. Conf. Spat. Cogn. (2006) 365–380.
- [75] Vila J., Beccue B., Anandikar S., The gender factor in virtual reality navigation and wayfinding, in: Proc. 36th Annu. Hawaii Int. Conf. Syst. Sci. HICSS 2003, IEEE, 2003: p. 7 pp. Doi: 10.1109/HICSS.2003.1174239.
- [76] S. Lee, Understanding Wayfinding for the Elderly Using VR, Virtual-Reality Contin. Its Appl. Ind. (2010) 285–288.
- [77] D. Head, M. Isom, Age effects on wayfinding and route learning skills, Behav. Brain Res. (2010), https://doi.org/10.1016/j.bbr.2010.01.012.
- [78] M. Kinateder, B. Comunale, W.H. Warren, Exit choice in an emergency evacuation scenario is influenced by exit familiarity and neighbor behavior, Saf. Sci. 106 (2018) 170–175, https://doi.org/10.1016/j.ssci.2018.03.015.
- [79] J. Lin, L. Cao, N. Li, How the completeness of spatial knowledge influences the evacuation behavior of passengers in metro stations: a VR-based experimental study, Autom. Constr. 113 (2020), 103136, https://doi.org/10.1016/j. autcon.2020.103136.
- [80] M. Fu, R. Liu, Y. Zhang, Why do people make risky decisions during a fire evacuation? Study on the effect of smoke level, individual risk preference, and neighbor behavior, Saf. Sci. 140 (2021), 105245, https://doi.org/10.1016/j. ssci.2021.105245.
- [81] M. Kinateder, M. Müller, M. Jost, A. Mühlberger, P. Pauli, Social influence in a virtual tunnel fire - Influence of conflicting information on evacuation behavior, Appl. Ergon. J. 45 (2014) 1649–1659, https://doi.org/10.1016/j. apergo.2014.05.014.
- [82] R.A. Ruddle, S.J. Payne, D.M. Jones, Navigating large-scale virtual environments: what differences occur between helmet-mounted and desk-top displays? Presence Teleoperators Virtual Environ. 8 (1999) 157–168, https://doi.org/10.1162/ 105474699566143.
- [83] P. Kiefer, I. Giannopoulos, M. Raubal, A. Duchowski, Eye tracking for spatial research: cognition, computation, challenges, Spat. Cogn. Comput. 17 (2017) 1–19, https://doi.org/10.1080/13875868.2016.1254634.
- [84] M. Tang, Analysis of signage using eye-tracking technology, Interdiscip. J. Signage Wayfinding. 4 (2020), https://doi.org/10.15763/issn.2470-9670.2020. v4.i1.a56.
- [85] C. Lander, N. Herbig, M. Löchtefeld, F. Wiehr, A. Krüger, Inferring landmarks for pedestrian navigation from mobile eye-tracking data and Google Street View, Conf. Hum. Factors Comput. Syst. - Proc. (2017) 2721–2729, https://doi.org/ 10.1145/3027063.3053201.
- [86] C. Ohm, M. Müller, B. Ludwig, Evaluating indoor pedestrian navigation interfaces using mobile eye tracking, Spat. Cogn. Comput. 17 (2017) 89–120, https://doi. org/10.1080/13875868.2016.1219913.
- [87] R.A. Schuchard, B.R. Connell, P. Griffiths, An environmental investigation of wayfinding in a nursing home, Eye Track. Res. Appl. Symp. 2005 (2005) 33, https://doi.org/10.1145/1117309.1117317.
- [88] P. Viaene, P. Vansteenkiste, M. Lenoir, A. De Wulf, P. De Maeyer, Examining the validity of the total dwell time of eye fixations to identify landmarks in a building, J. Eye Mov. Res. 9 (2016) 1–11, https://doi.org/10.16910/jemr.9.3.4.
- [89] Y.H. Bae, Y.C. Kim, R.S. Oh, J.Y. Son, W.H. Hong, J.H. Choi, Gaze point in the evacuation drills: analysis of eye movement at the indoor wayfinding, Sustain. 12 (2020) 1–14, https://doi.org/10.3390/su12072902.
- [90] H. Schrom-Feiertag, V. Settgast, S. Seer, Evaluation of indoor guidance systems using eye tracking in an immersive virtual environment, Spat. Cogn. Comput. 17 (2017) 163–183, https://doi.org/10.1080/13875868.2016.1228654.
- [91] D. Reid, Virtual reality and the person-environment experience, CyberPsychology Behav. 5 (2003) 559–564, https://doi.org/10.1089/109493102321018204.
- [92] K.M. Stanney, R.R. Mourant, R.S. Kennedy, Human factors issues in virtual environments: a review of the literature, Presence. 7 (1998) 327–351, https:// doi.org/10.1162/105474698565767.
- [93] B.G. Witmer, M.J. Singer, Measuring presence in virtual environments: a presence questionnaire, Presence Teleoperators Virtual Environ. 7 (1998) 225–240, https://doi.org/10.1145/985921.985934.
- [94] M. Kobes, I. Helsloot, B. De Vries, J. Post, Exit choice, (pre-)movement time and (pre-)evacuation behaviour in hotel fire evacuation - Behavioural analysis and validation of the use of serious gaming in experimental research, Procedia Eng. 3 (2010) 37–51, https://doi.org/10.1016/j.proeng.2010.07.006.
- [95] Patrick Costello, Health and safety issues associated with virtual reality a review of current literature, Advis Gr. Comput. Graph. (AGOGG) Tech. Rep. (1997) 1–23.
- [96] S. Schneider, K. Bengler, Virtually the same? Analysing pedestrian behaviour by means of virtual reality, Transp. Res. Part F Psychol. Behav. 68 (2020) 231–256, https://doi.org/10.1016/j.trf.2019.11.005.
- [97] C. Boletsis, J.E. Cedergren, VR locomotion in the new era of virtual reality: an empirical comparison of prevalent techniques, Adv. Human-Computer Interact. 2019 (2019), https://doi.org/10.1155/2019/7420781.
- [98] D. Rand, R. Kizony, U. Feintuch, N. Katz, N. Josman, A.S. Rizzo, P.L. Weiss, Comparison of two VR platforms for rehabilitation: video capture versus HMD, Presence. 14 (2005) 147–160, https://doi.org/10.1162/1054746053967012.
- [99] S. Sharples, S. Cobb, A. Moody, J.R. Wilson, Virtual reality induced symptoms and effects (VRISE): Comparison of head mounted display (HMD), desktop and

#### Y. Feng et al.

projection display systems, Displays. 29 (2008) 58-69, https://doi.org/10.1016/ j.displa.2007.09.005.

- [100] U. Dogu, F. Erkip, Spatial factors affecting wayfinding and orientation: a case study in a shopping mall, Environ. Behav. 32 (2000) 731–755, https://doi.org/ 10.1177/00139160021972775.
- [101] Y. Feng, D.C. Duives, S.P. Hoogendoorn, Development and evaluation of a VR research tool to study wayfinding behaviour in a multi-story building, Saf. Sci. 147 (2021), https://doi.org/10.1016/j.ssci.2021.105573.
- [102] J. Choi, E.R. Galea, W. Hong, Individual stair ascent and descent walk speeds measured in a Korean high-rise building, Fire Technol. 50 (2014) 267–295, https://doi.org/10.1007/s10694-013-0371-4.
- [103] K. Fitzpatrick, M.A. Brewer, S. Turner, Another look at pedestrian walking speed, Transp. Res. Rec. 2006 (1982) 21–29.
- [104] S. Mallaro, P. Rahimian, E.E. O'Neal, J.M. Plumert, J.K. Kearney, E.E. O'Neal, J. M. Plumert, S. Mallaro, P. Rahimian, A comparison of head-mounted displays vs. large-screen displays for an interactive pedestrian simulator, Proc. 23rd ACM Symp. Virtual Real. Softw. Technol. - VRST '17. (2017) 1–4, https://doi.org/ 10.1145/3139131.3139171.
- B.E. Riccke, B. Bodenheimer, T.P. McNamara, B. Williams, P. Peng, D. Feuereissen, Do we need to walk for effective virtual reality navigation? Physical rotations alone may suffice, Lect. Notes Comput. Sci. (Including Subser. Lect. Notes Artif. Intell. Lect. Notes Bioinformatics). 6222 LNAI (2010) 234–247, https://doi.org/10.1007/978-3-642-14749-4\_21.
- [106] E. Vilar, F. Rebelo, P. Noriega, L. Teixeira, E. Duarte, E. Filgueiras, Are emergency egress signs strong enough to overlap the influence of the environmental variables? in: A. Marcus (Ed.), Int. Conf. Des. User Exp Usability, Springer Berlin Heidelberg, Berlin, Heidelberg, 2013, pp. 205–214, https://doi.org/10.1007/ 978-3-642-39238-2 23.
- [107] N.A. Kaptein, J. Theeuwes, R. van der Horst, Driving simulator validity: some considerations, Transp. Res. Rec. (1996) 30–36, https://doi.org/10.3141/1550-05.
- [108] R.S. Kennedy, N.E. Lane, K.S. Berbaum, M.G. Lilienthal, Simulator sickness questionnaire: an enhanced method for quantifying simulator sickness, Int. J. Aviat. Psychol. 3 (1993) 203–220, https://doi.org/10.1207/s15327108ijap0303\_
- [109] J. Brooke, SUS A quick and dirty usability scale, Usability Eval. Ind. 189 (1996) 4–7, https://doi.org/10.1002/hbm.20701.
  [110] B.G. Witmer, C.J. Jerome, M.J. Singer, The factor structure of the presence
- [110] B.G. Witmer, C.J. Jerome, M.J. Singer, The factor structure of the presence questionnaire, Presence Teleoperators Virtual Environ. 14 (2005) 298–312.
- [111] F. Faul, E. Erdfelder, A. Buchner, A.G. Lang, Statistical power analyses using G\*Power 3.1: tests for correlation and regression analyses, Behav. Res. Methods. 41 (2009) 1149–1160, https://doi.org/10.3758/BRM.41.4.1149.
- [112] M. Gheisari. An Ambient Intelligent Environment for Accessing Building Information in Facility Management Operations, A Healthcare Facility Scenario, 2013.
- [113] J. Cohen. Statistical Power Analysis for the Behavioral Sciences, Academic press, 1988.
- [114] D. Paes, J. Irizarry, D. Pujoni, An evidence of cognitive benefits from immersive design review: comparing three-dimensional perception and presence between immersive and non-immersive virtual environments, Autom. Constr. 130 (2021), 103849, https://doi.org/10.1016/j.autcon.2021.103849.
- [115] G. Ozcelik, B. Becerik-Gerber, Benchmarking thermoception in virtual environments to physical environments for understanding human-building interactions, Adv. Eng. Informatics. 36 (2018) 254–263, https://doi.org/ 10.1016/j.aei.2018.04.008.
- [116] S. Jamieson, Likert scales: how to (ab)use them, Med. Educ. 38 (2004) 1217–1218, https://doi.org/10.1111/j.1365-2929.2004.02012.x.
- [117] S. Deb, D.W. Carruth, R. Sween, L. Strawderman, T.M. Garrison, Efficacy of virtual reality in pedestrian safety research, Appl. Ergon. 65 (2017) 449–460, https://doi.org/10.1016/j.apergo.2017.03.007.
- [118] S. Schwarzkopf, S.J. Büchner, C. Hölscher, L. Konieczny, Perspective tracking in the real world: Gaze angle analysis in a collaborative wayfinding task, Spat. Cogn. Comput. 17 (2017) 143–162, https://doi.org/10.1080/13875868.2016.1226841.

- [119] M. Haghani, M. Sarvi, Pedestrian crowd tactical-level decision making during emergency evacuations, J. Adv. Transp. 50 (2016) 1870–1895, https://doi.org/ 10.1002/atr.1434.
- [120] W. Liao, A. Seyfried, J. Zhang, M. Boltes, X. Zheng, Y. Zhao, Experimental study on pedestrian flow through wide bottleneck, Transp. Res. Procedia. 2 (2014) 26–33, https://doi.org/10.1016/j.trpro.2014.09.005.
- [121] D. Duives, H. Mahmassani, Exit choice decisions during pedestrian evacuations of buildings, Transp. Res. Rec. J Transp. Res. Board. 2316 (2012) 84–94, https://doi. org/10.3141/2316-10.
- [122] H. Li, J. Zhang, L. Xia, W. Song, N.W.F. Bode, Comparing the route-choice behavior of pedestrians around obstacles in a virtual experiment and a field study, Transp. Res. Part C Emerg. Technol. 107 (2019) 120–136, https://doi.org/ 10.1016/j.trc.2019.08.012.
- [123] M. Haghani, M. Sarvi, Human exit choice in crowded built environments: investigating underlying behavioural differences between normal egress and emergency evacuations, Fire Saf. J. 85 (2016) 1–9, https://doi.org/10.1016/j. firesaf.2016.07.003.
- [124] W. Liao, A.U.K. Wagoum, N.W.F. Bode, Route choice in pedestrians: determinants for initial choices and revising decisions, J. R. Soc. Interface. 14 (2017), https:// doi.org/10.1098/rsif.2016.0684.
- [125] P. Tian, Y. Wang, Y. Lu, Y. Zhang, X. Wang, Y. Wang. Behavior analysis of indoor escape route-finding based on head-mounted VR and eye tracking, IEEE Smart Data, IEEE, 2019, pp. 422–427, https://doi.org/10.1109/iThings/GreenCom/ CPSCom/SmartData.2019.00090.
- [126] Zielasko D., Weyers B., Bellgardt M., Pick S., Meibner A., Vierjahn T., Kuhlen T. W., Remain seated: Towards fully-immersive desktop VR, 2017 IEEE 3rd Work. Everyday Virtual Reality, WEVR 2017. (2017). Doi: 10.1109/ WEVR.2017.7957707.
- [127] S.J. Westerman, T. Cribbin, R. Wilson, Virtual information space navigation: evaluating the use of head tracking, Behav. Inf. Technol. 20 (2001) 419–426, https://doi.org/10.1080/01449290110069383.
- [128] E.M. Bourhim, A. Cherkaoui. Simulating pre-evacuation behavior in a virtual fire environment, IEEE, 2018, pp. 1–7, https://doi.org/10.1109/ ICCCNT.2018.8493658.
- [129] R. Kennedy, J. Drexler, D. Compton, K. Stanney, D. Lanham, D. Harm, Configural scoring of simulator sickness, cybersickness, and space adaptation syndrome: Similarities and differences, Virtual. Adam. Environ. Appl. Implic. Hum. Perform. Issues. (2003) 247–278, https://doi.org/10.1201/9781410608888.ch12.
- [130] Dominic J., Robb A., Exploring effects of screen-fixed and world-fixed annotation on navigation in virtual reality, in: 2020 IEEE Conf. Virtual Real. 3D User Interfaces Explor., 2020, pp. 607–615. Doi: 10.1109/VR46266.2020.00-21.
- [131] L. Rebenitsch, C. Owen, Review on cybersickness in applications and visual displays, Virtual Real. 20 (2016) 101–125, https://doi.org/10.1007/s10055-016-0285-9.
- [132] Reason J.T., Brand J.J., Motion sickness, Academic press, 1975.
- [133] W. Barfield, S. Weghorst, The sense of presence within virtual environments: a conceptual framework, Adv. Hum. Factors Ergon. 19 (1993) 699–704.
- [134] D. Nunez, How is presence in non-immersive, non-realistic virtual environments possible? ACM Int Conf. Comput. Graph. Virtual Real. Vis. Africa. 1 (2004) 83–86, https://doi.org/10.1145/1029949.1029964.
- [135] A. Bangor, P. Kortum, J. Miller, Determining what individual SUS scores mean: adding an adjective rating scale, J. Usability Stud. 4 (2009) 114–123.
- [136] J. Stigall, S. Sharma, Evaluation of mobile augmented reality application for building evacuation, Conf. Softw. Eng. Data Eng. Eval. (2019) 109–118.
- [137] F. Buttussi, L. Chittaro, Effects of different types of virtual reality display on presence and learning in a safety training scenario, IEEE Trans. Vis. Comput. Graph. 24 (2018) 1063–1076, https://doi.org/10.1109/TVCG.2017.2653117.
- [138] T. Hartmann, A. Trappey, Advanced Engineering Informatics Philosophical and methodological foundations with examples from civil and construction engineering, Dev. Built Environ. 4 (2020), 100020, https://doi.org/10.1016/j. dibe.2020.100020.